

# Pre-paid meters and household electricity use behaviours: evidence from Addis Ababa, Ethiopia

Abebe D. Beyene,<sup>1\*</sup> Marc Jeuland,<sup>2</sup> Samuel Sebsibie,<sup>1</sup> Sied Hassen,<sup>1</sup> Alemu Mekonnen,<sup>3</sup> Tensay Hadush Meles,<sup>4</sup> Subhrendu K. Pattanayak,<sup>2</sup> Thomas Klug<sup>2</sup>

<sup>1</sup>Environment and Climate Research Center, Policy Studies Institute, Addis Ababa, Ethiopia

<sup>2</sup> Sanford School of Public Policy, Duke University, Durham, USA

<sup>3</sup> Department of Economics, Addis Ababa University, Addis Ababa, Ethiopia

<sup>4</sup> School of Economics, University College Dublin, Belfield, Dublin 4, Ireland; Energy Institute, University College Dublin, Belfield, Dublin 4, Ireland







# Abstract

In low-income countries such as Ethiopia, pre-paid metering offers the potential to alleviate several typical challenges with traditional electricity billing systems, including high non-payment rate, pilferage and fraud, administrative and enforcement costs for utilities, and inflexibility and incongruence of bills with the irregular income flow of poorer consumers. Despite increasing adoption of pre-paid metering technology, few studies examine its impacts on household behaviour. This paper aims to fill this gap by examining impacts on electricity consumption, ownership of appliances, level of satisfaction, and cooking behaviour in Addis Ababa, the capital of Ethiopia. We employ propensity score matching and instrumental variable techniques to control for selection into pre-paid meters. Results indicate that pre-paid meter customers have significantly lower electricity consumption compared to post-paid users, and greater satisfaction with utility service. This technology also has a positive, but modest and statistically insignificant impact on total appliance ownership, and a positive and significant impact on ownership of energy-efficient lights. Furthermore, impacts are heterogeneous across customers: those who are more educated, who have higher income, and who do not share meters tend to reduce electricity use more. This evidence suggests the need for the utility to continue expanding the use of pre-paid meters and educating customers about their multiple advantages.

*Key words: Electricity utility; Energy access; Propensity score matching; Instrumental variables; Satisfaction* 

JEL: L94; O13; Q41

## Acknowledgements

This work was part of the research project 'Impacts and drivers of policies for electricity access: microand macroeconomic evidence from Ethiopia'. This project was funded with UK aid from the UK Government under the Applied Research Programme on Energy and Economic Growth (EEG), managed by Oxford Policy Management. We would lie also to thank participants of the Sixth Annual meeting of the Sustainable Energy Transitions Initiative (SETI), held virtually from June 17-19, 2021.

\*Corresponding author. Email: <u>abebed2002@yahoo.co.uk</u>

# 1 Introduction

Energy access is now prominent on the global development agenda, as reflected by its inclusion in the Sustainable Development Goals. However, recent attempts to increase access to electricity connections in low-income countries have created a set of new and complex challenges for utilities and consumers that collectively challenge sustainable development (McRae, 2015; Sievert and Steinbuks, 2020; Lukuyu *et al.*, 2021). Among these concerns is low revenue generation for the utility or other providers, which relates both to low electricity use among those newly connected and to low collection rates among electricity consumers (Lukuyu *et al.*, 2021; Blimpo and Cosgrove-Davies, 2019; Fobi *et al.*, 2018). Empirical evidence shows that high non-payment rates often force utilities to restrict electricity supply (Golumbeanu and Barnes, 2013). Another major challenge is electricity theft, which impedes revenue collection and hence infrastructure maintenance (Blimpo and Cosgrove-Davies, 2019; Smith, 2004). In addition, the traditional, post-paid billing system is costly to maintain and makes it difficult to reduce pilferage and fraud (Tewari and Shah, 2003). For cash-constrained consumers, meanwhile, monthly bills are inflexible and incongruent with the typically irregular nature of their income flows (Jack and Smith, 2015).

Pre-paid metering is increasingly deployed in both the electricity and water sectors as an innovative approach to address problems of non-payment, as well as to remove the mismatch between consumer access to cash and consumption (Heymans *et al.*, 2014; Jack and Smith, 2020). In sub-Saharan Africa (SSA) specifically, market forecasts suggest the greatest growth in electricity metering will come from pre-paid meters and that, by the end of 2034, the market value of prepaid electricity meters for SSA will grow by 234% (Northeast Group, 2014). South Africa was the first African country to introduce pre-paid meters in 1988, followed by Mozambique and Rwanda and, more recently, by Ethiopia, Nigeria, Angola, and Uganda (Esteves *et al.*, 2016). In Ethiopia, the location of our study, the Ethiopia Electric Utility (EEU) Company, a single and state-owned utility, has faced persistent challenges relating to billing collection, customer complaints (Walta Media and Communications Corporate SC, 2021), and calls for aggressive expansion of access to electricity (MoWIE, 2019).<sup>1</sup> As part of the technological solution to these challenges, the utility is increasingly replacing post-paid meters with pre-paid alternatives. Both utilities and policymakers in Ethiopia consider the installation of pre-paid meters to be one of the most critical tools for enhancing cost recovery (Tesfamichael *et al.*, 2021).

Despite growing deployment and adoption of this technology, relatively few studies have rigorously examined its impacts on household electricity consumption and other outcomes, however, especially in developing countries (Jack and Smith, 2015; Jack and Smith, 2020). This paper aims to fill this gap, examining the impact of the introduction of pre-paid metering in Addis Ababa, Ethiopia. We use quasi-experimental methods and aim to address several related questions: what is the impact of pre-paid metering on households' electricity consumption? How does this system affect other related variables, such as appliance ownership and in particular cooking technology alternatives and energy-efficient devices? Is pre-paid metering related to customer satisfaction with the services provided by the utility?

<sup>&</sup>lt;sup>1</sup> Due to problems collecting receivables from its customers, the EEU faces deficits. This has been responsible for delaying grid development and expansion of access to electricity for the remaining unconnected Ethiopian consumers. The EEU estimates these deficits currently amount to nearly US \$100 million per year: <u>https://www.rti.org/impact/expanding-electricity-services-and-energy-access-in-ethiopia</u>.

The rest of the paper is organised as follows. The next section (Section 2) presents a brief review of the literature. Section 3 focuses on discussing the background on electric meter replacement in Ethiopia. Section 4 presents the sampling technique, the nature of the data, and descriptive statistics of selected variables. Section 5 discusses the empirical strategies. Section 6 presents the estimation results, and the last section (Section 7) concludes.

# 2 Brief literature review

The shift to pre-paid billing is expected to have various impacts on electric consumers. Some of these impacts relate to behavioural changes undertaken by consumers as a result of pre-payment, given the greater flexibility of this system (Arthur *et al.*, 2010; Jack and Smith, 2015). Pre-payment also provides real-time information feedback about electricity consumption and its major drivers, which may increase consumer attention (Qiu *et al.*, 2017). Previous work has documented considerable advantages of pre-paid metering for utilities, including higher revenue collection (Jack and Smith, 2020; Tewari and Shah, 2003; Trimble *et al.*, 2016); reduction of non-technical losses, such as illicit connection and electricity theft (Kambule *et al.*, 2018; Mwaura, 2012); improved customer service and satisfaction (Mwangia and Mangusho, 2017; O'Sullivan *et al.*, 2004; Tewari and Shah, 2003); increased debt recovery (Tewari and Shah, 2003); reduction in disconnection and reconnection fees; and cash flow benefits from upfront payment (Tewari and Shah, 2003; The Allen Consulting Group, 2009).

Pre-paid metering also appears to induce energy saving, which can reduce pressure on limited transmission and generation capacity (Baptista, 2015; Jack and Smith, 2020; Kambule et al., 2018; Padam et al., 2018; Qiu et al., 2017). There is considerable evidence from high income countries on the general link between information feedback of the type provided by pre-paid metering and energy saving (Aydin et al., 2018; Blasch et al., 2019; Darby, 2001; Lynham et al., 2016; Ramos et al., 2015). This literature highlights the heterogeneity of impacts across different circumstances, typologies of information feedback, and socioeconomic settings. Pre-paid metering, as one of the typologies providing more direct and continuous information (Darby, 2001), can play a particularly strong role in reducing energy consumption by offering uniquely relevant information to households on how their electricity consumption varies with the use of different appliances (Ayodele et al., 2017; Arawomo, 2017). Empirical evidence supports this prediction. For example, following a randomised phase-in of new prepaid customers and observing consumption for four and a half years, Jack and Smith (2020) found a 14% reduction in electricity consumption in a sample from South Africa, which suggests such meters helped customers better understand and control electricity usage. Similar reductions have been observed in other developing country settings, e.g. among those receiving electricity consumption information in China (Du et al., 2017) and in Nigeria, where a study by Arawomo (2017) compared consumption data provided by meter readers and pre-paid meters and where a study by Ayodele et al. (2017) examined data from pre-paid meters alone.

This evidence notwithstanding, there is growing scepticism among researchers about the impact of prepaid metering, particularly on low-income households. For example, this payment system has been criticised for effectively hiding the difficulties low-income households face due to disconnection of service (O'Sullivan *et al.*, 2016; O'Sullivan *et al.*, 2014; Colton, 2001). O'Sullivan *et al.* (2014) argue that a pre-payment system forces vulnerable households to engage in a 'dichotomous choice between selfrationing and self-disconnection'. Another study in Tanzania argues that cash-constrained prepaid meter users may tend to rely on biomass fuels in order to avoid sudden disconnection (Jacome and Ray, 2018).

We make three main contributions in this paper. First, in deploying a quasi-experimental evaluation approach to control for non-random selection into connection with pre-paid meters, we add to a small handful of rigorous, empirical evaluations of the impacts of pre-paid metering in low-income countries. As discussed above, most quantitative studies on metering reforms address smart meters in a developed country context; studies in the developing world tend to be qualitative in nature or pertain to middleincome developing economies. The small set of studies relating to pre-paid metering in Ethiopia (Akele, 2012; Getachew, 2018) focus solely on the management and service quality pertaining to the introduction of a pre-paid metering system. Second, we leverage a unique and rich household dataset to examine a broader set of impacts than those on electricity consumption alone. This allows us to disentangle how consumption savings come about and shed light on several hypothesised negative effects of pre-paid meters. In particular, we are able to determine whether appliance ownership patterns change, specifically considering the balance of domestic labour-saving appliances (e.g. refrigerators, irons, washing machines, cooking appliances), entertainment devices (e.g. TVs, radios), and lighting technologies (e.g. rechargeable batteries, CFL or LED bulbs).<sup>2</sup> We also examine the impact of pre-paid metering on customer satisfaction of those serviced by the EEU. Third, we consider the possibility of heterogeneous impacts on different households, which further elucidates the equity implications of such a payment system.

Finally, beyond these specific contributions, our paper is timely and relevant for the policy environment in Ethiopia. The EEU is currently investing in expansion of pre-paid metering technology. As such, this study will help facilitate the formulation and implementation of evidence-based policy and in identifying the possibilities for further improvements in its implementation. The study not only provides evidence on the role of pre-paid metering as a demand side management tool, but also contributes to a more robust dialogue on how electricity billing and payment modalities in developing countries affect consumer well-being and energy transition.

# 3 Status of electric meter replacement in Ethiopia

The EEU, the state-owned electric power distribution agency in Ethiopia, is responsible for the distribution and sale of all of the country's grid electricity. The EEU operates in 11 regions and 28 districts and through nearly 560 customer service centres (CSCs). Despite its considerable problems, post-paid billing was the company's standard method of bill collection until 2007, prior to the advent of new metering technology. This traditional billing system, which remains in place in most of the country today, creates numerous challenges for the EEU. First, it requires lengthy and inefficient revenue collection procedures in that meter reading must be completed by utility personnel making the rounds of neighbourhoods and towns, followed by the generation and delivery of customer bills and finally payment collection. Second, non-payment and late payment for the electricity service are common, and the company incurs a high cost for legal enforcement (i.e., collection of penalties and eventual disconnection). Third, this billing system is prone to a range of errors, which can lead to inappropriately

<sup>&</sup>lt;sup>2</sup> Other appliances, such as sewing machines, solar lanterns, fans, etc., are owned by only a small proportion of the sample (less than 1% each) and hence are ignored in this analysis.

high or low billing amounts (Akele, 2012). For these and other reasons, post-paid billing has long been considered problematic by the utility, and attempts to improve on billing and revenue collection remain a major challenge.

In response to these problems and to better manage electricity supply and demand, the EEU began experimenting with a pre-paid metering system since around 2008, hoping to improve service quality in the process (Akele, 2012; Esteves *et al.*, 2016; Getachew, 2018). The dissemination of pre-paid metering began with a pilot project in the *Gerji* area of the capital city, Addis Ababa, implemented in 2007 in collaboration with an Egyptian meter manufacturer. Encouraged by the success of this pilot project, the EEU began disseminating these meters to its customers in 2008 (Getachew, 2018). At present, out of a total of more than 2 million domestic customers, about half a million have pre-paid meters.<sup>3</sup> The company is still rolling out the intervention and has plans to gradually switch all existing residential and non-residential meters to a pre-payment system (Akele, 2012). In Ethiopia, EEU believes that these meters will reduce non-technical losses, improve understanding of energy use and facilitate planning, help overcome revenue recovery and administration challenges, reduce customer debt, and facilitate improvement of customer service quality (Getachew, 2018).

Pre-paid meters also include in-home displays that provide information to consumers. Customers top up their account by buying a fixed amount of electricity from a nearby payment centre. When there is a disconnection (almost always because consumers fail to top up their account, rather than because of voltage fluctuation), pre-paid customers must travel to the nearest centre to pay to reconnect their meter. It is up to the customer to recharge their account if they want to continue drawing electricity from the grid. There is no penalty for exhausting the balance. Meters give a warning (blinking red lights) when a customer's balance drops below 30 birr.<sup>4</sup> In addition, the meter allows for a small amount of consumption on credit even after the balance drops to zero. This feature gives a customer time to replenish their account and helps reduce the inconvenience arising from sudden disconnection.

Currently, all domestic customers in Ethiopia are eligible for meter replacement, but meter replacement has been implemented gradually over time. New customers or existing post-paid users who apply for a new meter today are automatically assigned to the pre-payment system. Some households apply for meter replacement because they want to switch to pre-paid meters. In other cases, households may request meter replacement due to technical problems with the old post-paid meter. The third group of pre-paid meter users is composed of those moving into new residences, including condominium areas and newly built houses. These various aspects of selection for new meters inform the empirical strategy discussed in Section 5.

# 4 Sampling and data

### 4.1 Sampling

The sample for this study leverages the household Multi-Tier Framework (MTF) survey in Ethiopia administered by the World Bank as part of an international effort to better understand energy access in

<sup>&</sup>lt;sup>3</sup> This information was retrieved from the Addis Fortune newspaper: https://addisfortune.news/electric-billing-gets-worse-before-it-gets-better/

<sup>&</sup>lt;sup>4</sup> The Birr is the Ethiopian currency, with an exchange rate of US  $1 \approx 29$  birr at the time of the survey (August 2019).

low- and middle-income countries.<sup>5</sup> The MTF survey was designed to provide a nationally representative survey covering both urban and rural households. We draw on the sample from the first round to conduct a second-round survey focusing entirely on the major urban enumeration areas included in the original survey.

This particular study is based on data collected in Addis Ababa, which is where meter replacements have been most extensive. The complete sample comprises 1,182 households.<sup>6</sup> To construct this sample, all households enrolled in the first-round MTF survey from Addis Ababa were included. More than 78% of the sampled households in Addis Ababa who were surveyed in 2016 were revisited in 2019. The remaining households (22% of the sample) could not be found, for reasons such as relocation. At the time of the 2016 survey, only 8.9% of households had pre-paid meters in Addis Ababa. These pre-paid users were also enrolled in 2019, but the sample size was deemed insufficient for assessing the impact of pre-paid meters on the various outcome indicators evaluated in this study.<sup>7</sup> To increase the sample size of the 'treatment' group, we also recruited an additional 400 households not enrolled in the first-round survey from a list of pre-paid meter customers. This strategy allowed us to obtain a sufficient number of observations to have confidence in estimates of the impact of pre-paid metering on household electricity consumption.

Specifically, these additional pre-paid meter customers were selected according to the following twostep procedure. First, we distributed the total sample of pre-paid customers (obtained from the EEU) into the four regions (North, South, East, and West) of Addis Ababa, and then selected one centre from each region using a simple random sampling method. Each region has, on average, nine centres, with a minimum of seven and a maximum of 10 centres per region. The total number of centres in Addis Ababa is 36. For the second stage, 100 households were randomly selected from the list of all residential prepaid meter customers in each of the selected centres in the four regions.

#### 4.2 Descriptive statistics

The descriptive statistics show that 89.3% of the sampled households have a meter on the premises, while the remaining 10.7% are unmetered (i.e. they get electricity from their neighbours). Among the metered households considered in our analysis (N = 1,030), 56% are post-paid meter users, and the remainder are pre-paid subscribers. Meter sharing is commonplace in Addis Ababa: of the total sample households used in this empirical study, more than 26% share meters with their neighbours. This substantiates previous findings, which show meter sharing is widespread in Africa, and particularly in Ethiopia (Kojima *et al.*, 2016; Meles, 2020).

<sup>&</sup>lt;sup>5</sup> Details on the sampling procedure adopted in the first round can be found in Padam et al. (2018).

<sup>&</sup>lt;sup>6</sup> After dropping households without electricity meters (shared households) and those with incomplete information, 1,030 households were used for analysis. The shared households were those without an electric meter and which did not pay their electricity consumption expenses to the EEU. Instead, they paid them to a landlord or meter owner. The form of payment could be inclusion in the house rent, payment of a fixed amount per month, or some other arrangement for sharing the monthly electricity consumption expense (Meles, 2020).

<sup>&</sup>lt;sup>7</sup> This also makes it impossible to adopt the difference-in-difference (DID) estimation method, which requires longitudinal data to measure the impact of the 'treatment' by the difference between pre-paid and post-paid customers in the before–after difference in electricity consumption.

Table 1 presents the demographic characteristics of survey respondents. As shown, 44% of households are female-headed.<sup>8</sup> At first glance, female-headed households appear less likely to have pre-paid meters than male-headed households. More than half of post-paid meter users and only 35% of pre-paid meter users are female-headed. Pre-paid meter users appear more likely to be currently married. In addition, pre-paid meter users tend to have a smaller household size, to have relatively younger respondents, and to have more years of schooling compared to post-paid meter users. The difference in terms of education may not be surprising, as those who are better educated may have a greater understanding of the benefits of the pre-paid meter system. Finally, more than 43% of post-paid meter users and about 62% of pre-paid customers live in their own home. This suggests that ownership of the dwelling may be one of the factors driving connection to the pre-paid system, which aligns with expectations given the way the new metering system is being rolled out, especially for newly constructed homes and neighbourhoods.

Description of variables	Total (N=1053)	Post- paid (N=598)	Pre- paid (N=455)	Difference Mean
	Mean (1)	Mean (2)	Mean (3)	(4=2-3)
Household size	4.80	4.90	4.68	0.22*
Presence of children under 5 (= 1)	0.30	0.25	0.36	-0.10**
Age of household head	51.38	54.47	47.32	7.15***
Sex of household head (= 1 if female)	0.44	0.50	0.36	0.14***
Years of schooling of head	7.52	6.43	8.94	-2.51***
<i>Per capita</i> monthly consumption expenditure (birr)	1910.5	1514.7	2430.7	-915.9***
Marital status of household head (= 1 if married, 0 otherwise)	0.56	0.47	0.68	-0.21***
Dwelling ownership (= 1 if household owns house)	0.51	0.44	0.61	-0.17***
Number of rooms of dwelling	2.63	2.66	2.61	0.05
Meter sharing (= 1 if household shares meter)	0.24	0.24	0.24	-0.00
Household head unemployed (= 1 if unemployed)	0.21	0.24	0.18	0.05
Household head wage employed (= 1)	0.30	0.24	0.38	-0.14***
Household head self-employed (= 1 if self- employed)	0.21	0.19	0.24	-0.04

### Table 1: Summary statistics of variables used in the analysis

<sup>&</sup>lt;sup>8</sup> Similarly, the report prepared based on the 2016 MTF data shows 39.6% of households are female-headed in urban areas and 12% of households are female-headed in rural areas (Padam *et al.*, 2018).

Household head's occupation other (= 1 if household head engaged in other types of employment)	0.28	0.33	0.20	0.13***
Walls of house made of wood and mud (= 1)	0.71	0.82	0.58	0.24***
Walls of house made of concrete (= 1)	0.25	0.15	0.38	-0.23***
Walls of house made of other (= 1)	0.04	0.04	0.04	-0.01
Distance of the HH from the nearest centre in km	5.37	4.40	6.65	-2.25***
Dwelling connected to grid after 2007 (= 1) <sup>a</sup>	0.47	0.19	0.82	-0.63***
Proportion of pre-paid meter owners in CSC'	0.20	0.17	0.25	-0.07***

Source: Authors' computation from survey, 2019; p < 0.1, p < 0.05, p < 0.01. CSC refers to customer service centres. The sample size for the variable 'dwelling connected to grid after 2007' are 511 and 441 for post-paid customers and pre-paid users respectively.

### 4.3 Electricity consumption and income

Data on various measurements of the outcome variables—such as electricity consumption and expenditures; the type and number of appliance stock, such as lighting, housekeeping, and entertainment appliances; the type of main cooking and baking stoves; the use of efficient bulbs; and other additional information on the cooking behaviour of households and their level of satisfaction with the utility services—were collected by asking respondents about these in the survey. Households were asked about their electricity consumption (in kilowatt hours (kWh)) and about the bill they had paid for the month preceding the survey period (July 2019). This elicitation, based on recall, has several drawbacks. One is that the electricity consumption in the month immediately preceding the survey may not be a good representation of a household's actual electricity consumption pattern, due to seasonal or other variations. More importantly, obtaining an exact figure for electricity consumption from a survey is difficult for both post-paid and pre-paid customers, although the issues are distinct in each case. For pre-paid meter users, no bill is ever issued by the utility company. Instead, these consumers top up their meter whenever their account is out of credit. We calculated the monthly amount of electricity (kWh) consumed from the amount of money spent to top up their meter in the prior month, after deducting monthly service charges. Another challenge relating to pre-paid meters is that households do not top up at regular intervals, which makes it difficult to measure their exact monthly consumption. For post-paid meter users, bills are issued on a monthly basis but are subject to billing errors and complexities in amounts charged due to non-payment or arrears.

To address these issues, we complemented the survey data with billing data from the EEU covering the prior five years (2014-2018). The billing data for pre-paid customers were obtained from records based on actual recharge amounts purchased and were converted into monthly consumption amounts given the differing time intervals of such payments. Note we were only able to get such billing data for about 785 of the total sampled households (N = 1,030).<sup>9</sup> We then calculated the monthly average electricity

<sup>&</sup>lt;sup>9</sup> The remaining households did not match due to various inconsistencies in the survey and billing databases, especially relating to meter numbers. In cases where meter numbers did not match, we tried to match households based on the subscriber name in the billing data, but this was not always consistent either, as was the case for many renters or extended families. We have

consumption (kWh) and electricity spending from the billing data. The monthly averages were calculated from the total annual electricity consumption, accounting for any differences in the number of days covered by the bills. The average treatment effects were then estimated using monthly average electricity consumption as our main outcome variable. In addition, monthly electricity expenditure was used as a check on the robustness of our results.

Table 2 presents a summary of the key variables relating to electricity consumption and income. The average electricity consumption per month was 258 kWh,<sup>10</sup> higher than the average electricity consumption of 193 kWh per month found by the MTF survey for this particular subsample in 2016 (Padam *et al.*, 2018). Similarly, average household spending on electricity for 2019 was 245.5 birr (US\$ 8.5) per month, considerably higher than what was reported in 2016 for urban Ethiopia, 73.9 birr (US\$3.384) per month (Padam *et al.*, 2018). This increase in electricity spending is partly due to an increase in electricity consumption, and partly to a substantial upward tariff revision implemented in January 2019 that affected all consumers (pre-paid and post-paid). Monthly electricity consumption for pre-paid meter users was lower than that of post-paid meter users, and this difference is statistically significant. Similarly, a simple means comparison reveals that pre-paid meter customers' average monthly electricity spending was 44.2 birr (US\$ 1.5) lower than that of post-paid meter customers. Pre-paid meter users tend to have higher average monthly *per capita* expenditure and income compared to post-paid meter customers, however. This is in line with other literature, which finds a positive correlation between income and using pre-paid meters.

examined the distribution of missing data across the pre-paid and post-paid group and found it to be 29.6% and 22.9% respectively (Appendix 3A).

<sup>&</sup>lt;sup>10</sup> The average electricity consumption is almost the same (256.2 kWh/month) when we exclude the 400 additional pre-paid households enrolled since the first round of the MTF survey in 2016. This suggests that those additional households do not have systematically different electricity consumption on average relative to the original sample.

Table 2:	Summary	statistics for	key outcome	variables
----------	---------	----------------	-------------	-----------

	Full sample(1)		Post- paid(2)		Pre- paid(3)		Difference (2)-(3)	
	N	Mean	N	Mean	N	Mean	Mean	
Average monthly bill (birr)	1033	244.83	585	255.93	448	230.34	25.59**	
Average monthly electricity consumption (kWh)	1031	234.74	583	240.74	448	226.92	13.82	
Average monthly bill (based on utility data)	783	269.91	462	262.15	321	281.08	-18.93	
Average monthly electricity consumption (kWh) (based on utility data)	783	265.97	462	262.15	321	271.45	-9.30	
Appliance stock index	1053	0.93	598	0.82	455	1.06	-0.24***	
Lighting appliance index	1053	-0.73	598	-0.64	455	-0.84	0.19*	
Entertainment appliance index	1053	1.09	598	1.12	455	1.06	0.06**	
Housekeeping appliance index	1053	0.34	598	0.33	455	0.36	-0.04**	
Main bulb in house is energy-efficient	1044	0.83	593	0.82	451	0.86	-0.04**	
Main cookstove is electric (=1)	1053	0.73	598	0.71	455	0.76	-0.05	
Main baking stove is electric (=1)	1053	0.84	598	0.83	455	0.84	-0.01	
Frequency of cooking (days per week)	1053	6.58	598	6.55	455	6.62	-0.07	
Frequency of baking (days per week)	1053	1.74	598	1.75	455	1.72	0.03	
Household satisfaction with the utility	1055	0.67	599	0.65	456	0.70	-0.04	

Source: Authors' computation from survey, 2019. \*, \*\*, and \*\*\* indicate statistical significance at 10%, 5%, and 1%, respectively. The indices are constructed based on polychoric Principal Component Analysis.

# 5 Theoretical framework and estimation strategy

## 5.1 Conceptual framework

Pre-paid metering is believed to influence energy consumption via nudging, price effects, information provision, and costs of discontinuation (Qiu *et al.*, 2017). First, the information feedback channel arising from in-home displays and real-time feedback can encourage energy conservation. Frequent reminders about energy consumption among those who pay attention to the displays provide nudges that aid budgeting and are generally believed to reduce consumption, at least for those who are resource-constrained or conservation-minded. Second, information provided by pre-paid meters is more likely to be salient. In our context, even inattentive consumers receive reminders when their balance is depleted. Households are further incentivised to pay attention to their electricity consumption because not monitoring their balance entails significant inconvenience relating to disconnection. Thus, unless the recharge amounts are large and infrequent, it seems likely that consumers obtain additional and more salient information with this setup than they would with the typical post-paid system. Third, the prepaid program reduce consumption as future payments in the case of post-paid mechanism provide additional

advantage due to discounting. However, this effect tends to be insignificant as the time interval is too short and the interest rate is small. Fourth, failure to top up a pre-paid account creates additional costs for the households relating to disconnection. For those who remain inattentive despite the additional information feedback, these disruptions will increase the relative cost of electricity, inducing both income and substitution effects. The latter may also result from the upfront nature of payments under the pre-paid system, which must be compared with the discounted future benefits of that energy consumption.<sup>11</sup>

Consistent with this framework, existing research has found that in-home-displays alone can induce energy conservation (Stinson *et al.*, 2015). Still, this impact may vary across individuals and the way displays are designed, having been shown to depend on various factors like message framing (Schultz *et al.*, 2015), the initial level and type of energy consumption (Matsukawa, 2018), and characteristics of the target population (Aydin *et al.*, 2018; Krishnamurti *et al.*, 2013). On the other hand, the pre-payment mechanism is expected to trigger behavioural change among electricity consumers due to changes in the timing of payment (Malama *et al.*, 2014). For example, studies have observed an improvement in household budgeting, with many households also reporting that they ration credit when it is low, reduce electricity use and debt, and stop using electricity for energy-intensive activities like cooking (Malama *et al.*, 2014). In our setting, information and payment timing mechanisms are relevant but cannot be identified separately, since all pre-paid customers also have access to in-house displays that provide feedback on energy consumption. Based on prior evidence, however, we expect the pre-paid billing system will tend to reduce electricity consumption.

We also examine the heterogeneous impacts based on different factors such as education, income, and meter sharing. We expect impacts would be greater, *ceteris paribus*, for those at the lower end of the income distribution, those who do not share meters, and those who are more educated (and perhaps understand information better). We also hypothesise an impact of pre-paid metering on the ownership of appliances and on the use of electric stoves since these consumers are perhaps more conscious of their consumption patterns and therefore better able to control their electricity use, as well as positive effects on customer satisfaction due to the enhanced reliability of their electricity service and on the use of energy-efficient bulbs that conserve energy.

### 5.2 Model specification

Our empirical analysis of the impact of pre-paid meters on various outcomes involves estimation of average treatment effects, accounting both for selection on observables and for selection on unobservables. To adjust for the former, we estimate the average treatment effect on the treated (ATT) using propensity score matching (PSM). To also account for the latter, we further implement an instrumental variable (IV) estimation approach.

Our primary interest is to estimate the average treatment effect of having a pre-paid meter on electricity consumption, satisfaction with electricity service, and ownership and use of various appliances, including electric stoves and energy-efficient light bulbs. The appliance category is also

<sup>&</sup>lt;sup>11</sup> In our case, the period over which discounting occurred was relatively short, but previous research also shows that individuals tend to discount outcomes in the near future more compared to outcomes in the distant future, and that discount rates among those facing tight income constraints can be high (Frederick *et al.*, 2002). This will reinforce income and substitution effects.

disaggregated into cooking, entertainment, lighting, and domestic labour-saving equipment groups. The treatment takes a binary value of  $T_i = 1$  if household i is enrolled in the pre-payment programme, and  $T_i = 0$  if household i is a post-paid customer. Let  $Y_i$  denote the outcome variable (e.g. monthly electricity consumption) of household i. This variable  $Y_i$  takes values of  $Y_1$  and  $Y_0$  for pre-paid (treatment) and post-paid (control) users respectively. Following Rosenbaum and Rubin (1983), the average treatment effect (ATE) is then defined as

$$ATE = E(Y_1 | T_i = 1) - E(Y_0 | T_i = 0)$$
(1)

Direct estimation of the above equation is problematic as the two groups are not directly comparable owing to the selection by various means (discussed in Section 4) into the pre-paid metering group. In such cases, the average difference in the outcome after treatment cannot be reasonably attributed to the treatment, since observable and unobservable pre-treatment differences may be partly responsible for the change in the outcome of interest. To address those problems, we apply PSM and IV methods.

### 5.2.1 Propensity Score Matching (PSM)

In this approach, we assume that assignment into pre-paid metering is driven only by observed characteristics, in which case knowledge of the vector of these observed variables  $X_i$  is sufficient to identify the unbiased impact of the treatment on the outcome. Under this assumption, we estimate the average treatment effects on the treated using PSM based on observables. The idea is to create a control group that is statistically comparable to the pre-paid metering group and that approximates the counterfactual shown in Equation (1), because it is observationally comparable with the treatment group in terms of characteristics that are unaffected by the treatment (Caliendo and Kopeinig, 2008; Becker and Ichine, 2002). According to Rosenbaum and Rubin (1983), the propensity score, Pr ( $T_i = 1/X$ ), is defined as the propensity of exposure to a particular treatment, given a set of observed covariates. Mathematically, the conditional probability that a household has a pre-paid meter, conditional on observed characteristics X, is given by:

$$P(X) = \Pr\left(T_i = 1 | X_i\right) \tag{2}$$

The validity of matching on propensity scores depends on two basic assumptions: i.e. *the overlap condition*, which requires that a sufficient number of observations in the treatment have corresponding untreated comparisons based on their propensity scores; and the *unconfoundedness* condition (also called conditional independence), which requires that, given the observed covariates X, outcomes are independent of the treatment assignment (Khandker *et al.*, 2010). If these assumptions are satisfied, the treatment assignment is said to be strongly ignorable, given the observed covariates X and given the propensity score P(X). Under strong ignorability, units with the same propensity score but with a different treatment condition, can be used as treatment and control (Rosenbaum and Rubin, 1983), and the ATT is:

$$ATT = E(Y_1|T_i = 1, P(X)) - E(Y_0|T_i = 1, P(X))$$
(1)

Our PSM analysis was implemented in Stata using the *psmatch2* module. Propensity scores were first obtained for each of 1,029 observations for which the survey data were complete and which had electricity connections (as discussed in Section 3), and then for the restricted sample with verifiable billing data. Table 3 reports the results of a logit specification used to estimate the probability shown in

Equation (2) (columns 1 and 2). It displays the determinants of selection into having a pre-paid meter as a function of a number of household variables we believe are likely to affect the likelihood of using prepaid metering technology.<sup>12</sup> Column 1 is for the complete sample, while column 2 is for the subsample for which we are able to match billing data. The signs and levels of significance of the variables in the first and second columns are generally consistent. Thus, these determinants do not appear sensitive to restricting the analysis to the sample for which we are able to verify billing data. Column 3 and 4 also show the first stage regression results for the IV specification that is described in further detail below. Most of the results are qualitatively similar. The results show that older people are less likely to use prepaid meters, while married households are more likely to have them. Unlike the findings of Oseni (2015), households with high *per capita* consumption expenditure are more likely to have pre-paid meters, as are households who own their dwelling or whose dwellings have more rooms. Distance from the nearest utility centre is found to be positively and significantly associated with the probability of using a prepaid meter, which is intuitive regarding differential convenience of bill payment, as well as the manner in which pre-paid meters are rolled out (with greater use in newer and therefore outlying housing developments).

	Logit for PSM- Entire sample	Logit for PSM- reduced sample	Probit for IV- entire sample	Probit for IV- reduced sample
Household size	-0.02	-0.03	0.02	0.03
	(0.04)	(0.05)	(0.03)	(0.04)
Presence of children under 5	0.17	0.21	-0.00	-0.00
Presence of cinturen under 5	(0.17)	(0.21)	(0.12)	(0.14)
Age of bood	-0.02***	-0.02**	0.00	0.00
Age of field	(0.01)	(0.01)	(0.00)	(0.01)
Soy of head	0.15	0.04	0.32**	0.23
Sex of field	(0.19)	(0.22)	(0.13)	(0.16)
Schooling of head	0.02	0.02	0.03**	0.03**
Schooling of head	(0.02)	(0.02)	(0.01)	(0.01)
Consumption expanditure (In)	0.33***	0.37***	0.16**	0.19**
	(0.11)	(0.13)	(0.07)	(0.09)
Marital Status	0.85***	0.89***	0.45***	0.50***
	(0.18)	(0.22)	(0.13)	(0.16)
Dwolling ownorchin	0.83***	1.02***	0.32***	0.32**
Dwetting ownership	(0.16)	(0.19)	(0.11)	(0.13)

# Table 3: First-stage regression estimates

<sup>&</sup>lt;sup>12</sup> We have checked the results by including regional dummies representing utility districts and found that most of the results are qualitatively similar but differ in level of significance which may be due to reduced degrees of freedom as a result of large number of explanatory variables.

Number of reems	-0.23***	-0.22***	-0.11***	-0.08
	(0.06)	(0.07)	(0.04)	(0.05)
Motor charing	-0.21	-0.13	0.15	0.24*
	(0.17)	(0.21)	(0.12)	(0.14)
Hoad is wage employed	0.29	0.55**	0.05	0.26
nead is wage employed	(0.22)	(0.27)	(0.16)	(0.19)
Hoad is solf amployed	0.07	0.15	-0.14	-0.17
neau is sett employed	(0.23)	(0.28)	(0.16)	(0.19)
Head amployed in other	-0.07	-0.05	0.01	0.02
head employed in other	(0.22)	(0.27)	(0.15)	(0.18)
Walls made of wood and mud	-0.43	-0.38	-0.08	0.01
watts made of wood and mud	(0.35)	(0.40)	(0.25)	(0.29)
Walls made of concrete	0.53	0.46	0.19	0.18
walls made of concrete	(0.37)	(0.43)	(0.26)	(0.31)
Distance from pearest centre	0.04***	0.06***	0.01**	0.02
	(0.01)	(0.02)	(0.01)	(0.01)
Grid-connected after 2007			1.67***	1.81***
ond-connected after 2007			(0.10)	(0.13)
Properties of pre-paid in CSC			0.81***	0.93***
Proportion of pre-paid in esc			(0.29)	(0.34)
cons	-2.30**	-2.97**	-3.00***	-3.60***
	(1.03)	(1.22)	(0.73)	(0.88)
Ν	1050	779	987	736

\*\*\*, \*\*, and \* indicate statistical significance at 1%, 5%, and 10%, respectively; standard errors in parentheses

Furthermore, as depicted in Figure 1, there exists a sizeable region of common support between the two groups in both the full sample and the confirmed billing data subsample. Observations outside the common support region were dropped from further analysis.



### Figure 1: Distribution of propensity scores

Note: 'Treated: on support' refers to observations in the pre-paid group that have a suitable comparison. 'Treated: off support' refers to the observations in the pre-paid group that do not have a suitable comparison.

To assess robustness across different specifications of the matching algorithm, we applied various commonly-used approaches: nearest-neighbour matching (NNM) with one, three, and five neighbours, kernel matching, and radius matching with three caliper levels (0.1, 0.05, and 0.01).<sup>13</sup> As suggested in the literature, there is no matching algorithm that dominates in all data situations and the choice of specific algorithms involves a trade-off between bias and efficiency (Caliendo and Kopeinig, 2008). In each case, following Rubin (2001), we estimate the standardised difference in the mean value (Rubin's B) and the variance ratio (Rubin's R) of the propensity score for pre-paid and post-paid meter users. Standardised differences greater than 20% are considered high (Rosenbaum and Rubin, 1985). The distributions of propensity scores (for the example of radius matching with a caliper level of 0.05) are shown in Figure 2. As shown, there is substantial reduction in the bias after matching. In other words, there is sufficient similarity on observables between households in the treatment group and those in the control group to allow appropriate comparisons, and the overlap condition is satisfied.<sup>14</sup>

<sup>&</sup>lt;sup>13</sup> In this paper, we apply matching with replacement instead of matching without replacement, as the former allows a better match to be used more than once and reduces bias (Dehejia and Wahba, 2002). The latter, on the other hand, may increase bias but might improve the precision of the estimates. As discussed in Rosenbaum (1996), the results obtained based on matching without replacement may be sensitive to the order in which the treatment units are matched.

<sup>&</sup>lt;sup>14</sup> As suggested in Rosenbaum and Rubin (1985), we examined the reduction in the mean absolute standardised difference between the pre-paid customers and post-paid customers before and after matching. The difference or bias is calculated for each covariate. In our case, the standardised mean difference for overall covariates used in the propensity score is less than 25% for most of the algorithms (Appendix 1). Furthermore, the p-values of the likelihood ratio tests show that the joint significance of the covariates was always rejected after matching in all matching algorithms. All the tests (low pseudo R2, low standardised bias, and high total bias reduction) suggest that PSM was successful in balancing the distribution of covariates

# Figure 2: Distribution of propensity scores before and after matching, for radius matching algorithm with caliper level of 0.05



Of course, if unobserved factors affect selection into the use of pre-paid meters, the strong ignorability assumption required for PSM no longer holds, and the results from PSM will be biased (Rosenbaum, 1996). Indeed, the assignment of pre-paid meters in Ethiopia is likely to be affected by an array of observed and unobserved factors. As discussed further in the results, we check the robustness of our estimated results across various methods. First, we tested for the presence of hidden bias drawing from sensitivity analysis, as proposed by Rosenbaum (1996, 2002, 2005). Results were generally insensitive to small bias (for instance bias that would alter the odds of treatment assignment by 20% or so), but could be sensitive to larger bias. Furthermore, we complemented our analysis with IV estimation of average treatment effects, as described below.

### 5.2.2 IV estimation of average treatment effects

Following Wooldridge (2010) and Cerulli (2015), the IV estimation of treatment effects is given as follows. Let

$$Y = Y_0 + T(Y_1 - Y_0)$$
  
$$Y = \mu_0 + T(\mu_1 - \mu_0) + \nu_0 + T(\nu_1 - \nu_0)$$
 (4)

where  $Y_0 = \mu_0 + \nu_0$ ,  $Y_1 = \mu_1 + \nu_1$  and  $\nu_0$  and  $\nu_1$  are random components of  $Y_0$  and  $Y_1$  respectively. We further assume that the weak version of the conditional mean independence assumption does not hold, i.e.:  $E(\nu_1|T, X) \neq E(\nu_1|X)$  and  $E(\nu_0|T, X) \neq E(\nu_0|X)$ 

between the two groups. We have undertaken these analyses for the various matching algorithms applied, and results are consistent across approaches. More details can be found in Appendix 1, and additional diagnostics are available upon request from the authors.

For parameter estimation, we further consider two scenarios. In the first scenario, we assume homogenous treatment effects, in which case Equation (4) becomes:

$$Y = \mu_0 + T(\mu_1 - \mu_0) + \nu_0 \tag{5}$$

Suppose there is a vector of instruments Z that satisfies the conditions that Z is uncorrelated with  $v_0(E(v_0|X,Z) = E(v_0|X))$  and correlated with the treatment  $(E(T|X,Z) \neq E(T|X))$ . Under those conditions, the outcome equation and the latent selection functions are given by:

$$Y_i = \mu_0 + T_i ATE + X\beta + u_i$$
 and  $T_i = \rho + q_i \delta + \varepsilon_i$  respectively,

where  $q_i = (X_i, Z_i)$  is an identification restriction and  $E(v_0|X, Z) = E(v_0|X) = g(X\beta)$ . The outcome equation cannot be consistently estimated due to the endogeneity of the treatment  $T_i$ . In such cases, IV regression estimated by probit and two-stage least squares provides consistent estimation of the outcome equation, irrespective of the correct specification of the selection model (Cerulli, 2015). The procedure is as follows. First, the predicted probability of treatment is estimated by estimating a probit model of the binary treatment as a function of a set of covariates X and instruments Z. Second, the predicted probabilities from the selection model are used to instrument for the treatment  $T_i$ , and twostage least squares regression is applied to estimate the average treatment effect (Cerulli, 2015). Under the assumption of homogenous treatment effects, the average treatment effect (ATE) and average treatment effect on the treated (ATET/ATT) are equal.

In the second scenario, we assume there is a heterogeneous response to the treatment, which depends on covariates *X*, such that the ATE and ATET/ATT are not equal. In this case, the outcome equation becomes:

$$Y_i = \mu_0 + T_i ATE + X\beta_0 + (X - \mu_X)\beta + \varepsilon_i$$
(6)

Equation (6) can also be consistently estimated by applying probit–2SLS (Cerulli, 2015).

In order to estimate the IV regression, valid instruments are needed that have a direct effect on the use of pre-paid meters and only affect electricity consumption indirectly via the use of pre-paid meters. In other words, the instruments must first be correlated with the endogenous variables (switching to a pre-paid meter) and, second, they must not be correlated with the other unobserved factors that affect electricity consumption.

We identified two potential instruments from the survey and billing data collected from EEU. The first is the proportion of pre-paid meter customers to total customers in a given CSC. If there are more customers with pre-paid meters in a given centre, it is highly likely that information spread about the technology will induce other customers in the centre to switch to that system.<sup>15</sup> Centre-level dissemination should, however, be exogenous to individual household characteristics, so households residing in a centre with more pre-paid customers are more exposed to the technology and become

<sup>&</sup>lt;sup>15</sup> The correlation coefficient between pre-paid meter ownership and the proportion of pre-paid meter customers in a centre is 0.665, but the correlation coefficient between household electricity consumption and the proportion of pre-paid meter customers in a centre is -0.017, supporting our argument that it can be a valid instrument.

better aware of it, and this differs from centre to centre.<sup>16</sup> We collected data on the number of pre-paid and post-paid meter customers in each of the 36 CSCs in Addis Ababa. The instrument at the centre level was then computed as follows:

 $Proportion of pre-paid customers = \frac{The number of pre-paid costomers}{Total number of customers}$ 

The second instrument employed is a variable representing whether the customer's house was connected to the national grid after 2007 or not. From the household survey, households were asked about the number of years their house had been connected to the national grid. The intuition behind this instrument is that the distribution of pre-paid meters in Addis Ababa started in 2007, and most newly built houses were automatically assigned to pre-paid electricity meters after that date. In addition, it is highly likely that older dwellings connected to the grid after 2007 were assigned to pre-paid electricity meters. Our first-stage regression result suggests the instruments are relevant, as they are significant at the 1% level (Table 3, column 3). Moreover, the signs and significance of variables in the IV probit generally appear similar to those for the first stage of the PSM model, with a few exceptions. This is expected, due to the inclusion of the two additional IVs (note that the IV estimation is only possible for the subsample with matched billing data, for which we can also identify the timing of the grid connection).

# 6 Estimation results

### 6.1 Average treatment effects using PSM

This section presents the effect of pre-paid metering on electricity consumption, appliance ownership, cooking behaviour, and household satisfaction with utility service, as derived from the PSM approach. The tables below present the average treatment effects on the treated (ATT), with bootstrapped standard errors. As shown, the use of pre-paid meters results in significantly lower electricity consumption. The estimation result shown in Table 4 shows that the effect ranges from 14% to 23%. Given that the outcome variable is in logarithmic form, prepaid customers have, on average, 13% to 20.5% lower electricity consumption per month compared to post-paid meter users. Alternatively, when the outcome is measured in terms of monthly spending on electricity, households with pre-paid meters have, on average, 19.7% to 26.6% lower monthly electricity spending compared to similar households with post-paid meters. Results show only minor differences<sup>17</sup> across the different matching algorithms (radius and kernel matching and NNM), which indicates results are robust to the specific PSM algorithm that is applied.

<sup>&</sup>lt;sup>16</sup> Community-level proportions of this kind have been used as instruments in the literature. For example, Jung and Streeter (2015) used the average take-up rate of insurance in the community as a valid instrument for the endogenous variable 'enrolment of health insurance'.

<sup>&</sup>lt;sup>17</sup> Note the dependent variables are log transformed.

# Table 4A: Estimates of average treatment effects: PSM results for the entire sample (using<br/>self-reported monthly electricity consumption)

Matching algorithm		Bill	kWh	App. Stock	Lighting App.	Entertain- ment App.	House- keeping App.
	Caliper=0.01	-0.27***	-0.18***	0.07	-0.12	-0.09**	-0.02
		(-0.07)	(-0.06)	(-0.06)	(-0.14)	(-0.04)	(-0.02)
Padius	Caliper=0.05	-0.23***	-0.15***	0.07	-0.09	-0.08**	-0.02
Raulus		(-0.06)	(-0.06)	(-0.06)	(-0.13)	(-0.04)	(-0.02)
	Caliper=0.1	-0.22***	-0.14**	0.08	-0.13	-0.08**	-0.01
		(-0.06)	(-0.05)	(-0.06)	(-0.13)	(-0.04)	(-0.02)
	NN=1	-0.31***	-0.23**	0.08	-0.15	-0.07	0
Nearest		(-0.09)	(-0.1)	(-0.09)	(-0.17)	(-0.06)	(-0.03)
neighbour	NN=3	-0.29***	-0.22***	0.06	-0.08	-0.09**	-0.02
(NN)		(-0.08)	(-0.09)	(-0.07)	(-0.15)	(-0.04)	(-0.02)
	NN=5	-0.26***	-0.20***	0.07	-0.1	-0.09*	-0.02
		(-0.07)	(-0.07)	(-0.08)	(-0.15)	(-0.05)	(-0.02)
Kernel	BW=0.06	-0.23***	-0.15**	0.06	-0.09	-0.08**	-0.02
		(-0.06)	(-0.06)	(-0.06)	(-0.12)	(-0.04)	(-0.02)
Ν		1025	1024	1050	1050	1050	1050

Standard errors are in parentheses; \*p< 0.10, \*\*p< 0.05, \*\*\*p< 0.01

# Table 4A: Continued

Matching Algorith	ms	Energy Eff. Bulb	Main cookstove	Main bakingstove	Cooking freq.	Baking freq	Satisfaction
	Caliper=0.01	0.05	0.00	0.02	0.02	-0.01	0.11***
		(0.04)	(0.04)	(0.04)	(0.10)	(0.06)	(0.04)
Padius	Caliper=0.05	0.05	0.01	0.03	0.00	-0.01	0.10**
Raulus		(0.03)	(0.03)	(0.04)	(0.09)	(0.06)	(0.04)
	Caliper=0.1	0.06*	0.01	0.03	0.00	-0.02	0.09**
		(0.03)	(0.03)	(0.04)	(0.09)	(0.06)	(0.04)
	NN=1	0.10**	0.02	0.02	-0.04	0.00	0.08
		(0.05)	(0.05)	(0.05)	(0.13)	(0.10)	(0.06)
NIN	NN=3	0.07	-0.00	0.03	-0.02	-0.02	0.08
ININ		(0.04)	(0.04)	(0.05)	(0.10)	(0.08)	(0.06)
	NN=5	0.07**	-0.01	0.02	-0.02	-0.02	0.10*
		(0.04)	(0.04)	(0.04)	(0.11)	(0.07)	(0.05)
Karnal	BW=0.06	0.05	0.00	0.03	0.00	-0.01	0.10**
Kernet		(0.03)	(0.03)	(0.03)	(0.09)	(0.07)	(0.04)
Ν		1041	1050	1050	1050	1050	1050

Standard errors are in parentheses; \*p< 0.10, \*\*p< 0.05, \*\*\*p< 0.01

Table (D. Tables to a fa

We conducted a similar exercise for the sample with actual billing data (Table 4B). The ATT estimates for this reduced matched sample are higher than those based on survey-reported electricity consumption from the previous month (Table 4A). Depending on the specific matching algorithm used, the estimated impact of pre-paid meters (measured in kWh) is estimated to range from 22 to 28%. This shows that, on average, prepaid adopters have 19.8% to 24.4% lower electricity consumption than non adopters. Similarly, the estimated impact on electricity expenditure (measured in Ethiopian birr) ranges from 21% to 27% showing that prepaid adopters have on average 19% to 23.7% lower monthly electricity spending than non adopters..<sup>18</sup> Results are significant and similar, irrespective of the type of matching algorithm used.

These basic findings on energy consumption are in line with other findings from similar studies. For example, Jack and Smith (2015) found that households with pre-paid meters in Cape Town had 14% lower consumption of electricity. In a developed world context, a 12% reduction was found in Arizona in the United States (Qiu *et al.*, 2017).

Table 4B:	Estimates of average treatment e	mects: PSM on matched sa	mple using metering
	information obtained from the bi	lling data from EEU	

and the standard offerster DCM and the standard second s

Matching Algorithms		Bill from EEU	kWh from EEU
	Caliper=0.01	-0.23**	-0.25***
Radius		(0.09)	(0.09)
	Caliper=0.05	-0.23**	-0.25***
		(0.09)	(0.09)
	Caliper=0.1	-0.21**	-0.22**
		(0.09)	(0.09)
	NN=1	-0.27***	-0.28***
		(0.10)	(0.10)
NINI	NN=3	-0.26***	-0.27***
		(0.09)	(0.09)
	NN=5	-0.26***	-0.27***
		(0.10)	(0.10)
Kornol	BW=0.06	-0.24**	-0.25***
Kennet		(0.09)	(0.09)
Ν		779	779

Standard errors in parentheses; \* p<0.10, \*\* p<0.05, \*\*\* p<0.01

Turning to the other outcome variables of interest, our first observation is that pre-paid metering has a positive, but modest and statistically imprecise influence on total appliance ownership. Further analysis

<sup>&</sup>lt;sup>18</sup> The results show that effects on expenditure are lower with actual bills, but higher with survey data. The main reason for this might be the load-shedding in the city just before the survey period (previous months), which may affect household consumption behaviour.

by appliance type does not reveal statistically significant impacts on specific categories of entertainment, domestic labour-saving, or lighting devices. Thus, the overall effect may be slightly positive on ownership across all categories, but with clearly positive and significant impact only on the use of energy-efficient light bulbs, which is encouraging, as the promotion of such energy-saving devices is also a goal of the EEU. Pre-paid meters do not appear to have any meaningful impact on household cooking behaviour, as reflected by the number of cooking and baking episodes per week, or on the use of electric stoves.

Some literature, mostly qualitative, finds a link between customer satisfaction and pre-paid metering (Mahapatra and Golhar, 2018; Mwangia, and Mangusho, 2017; O'Sullivan *et al.*, 2014; Chinomona and Sandada, 2014). Our analysis supports this interpretation. Depending on the matching methods, the level of satisfaction among pre-paid consumers is 9.0 to 11.0 percentage points higher than among post-paid customers. This suggests that, in addition to saving energy, pre-paid metering improves household perception of electricity service quality.

As discussed above, PSM reduces selection bias arising from observables but does not address unobservables, nor does it allow for a direct test of the influence of the latter. Ichino *et al.* (2008), therefore, suggest that estimation of causal effects through matching under the unconfoundedness assumption should be followed by sensitivity analysis to understand the degree of reliability of the derived estimates. Following Rosenbaum (2002), we computed the bounds test for sensitivity (see also DiPrete and Gangl (2004) for additional details). The associated values of  $\Gamma$  estimated using the *rbounds* test are presented in Appendix 4 for each estimate.<sup>19</sup> This sensitivity analysis indicates that our estimates are not sensitive to small biases but may be affected by moderate to large bias. To be specific, when the outcome is measured in terms of electricity consumption (kWh) per month, a hidden bias that alters the odds of treatment assignment by more than 15% could result in a change in our conclusion about the impacts of pre-paid meters. To further verify robustness of our estimates to such hidden bias, we supplement the matching analysis with the IV analysis (Section 6.3).

### 6.2 Heterogeneous impacts using PSM

The ATT estimates reported in Table 4A and 4B above assume homogeneous impacts of pre-paid meters, but these impacts may differ across different household types. For example, household electricity consumption behaviours, as well as their attention to new information, are variable, creating the possibility that pre-paid metering may also have differential impacts. Previous studies (e.g. Aydin *et al.*, 2018; Bao and Ho, 2015) reveal that information provision can have heterogeneous impacts that vary according to household characteristics. In this section, we present the heterogeneous treatment effect of pre-paid metering on our outcomes of interest, as a function of several characteristics (income, education, and meter sharing).

Heterogeneous impacts across income groups (Table 5A) were analysed based on a median income cutoff. We find that electricity consumption decreases for both low-income and high-income groups with pre-paid meters, but that the effect is greater for higher-income households. This is consistent with

<sup>&</sup>lt;sup>19</sup> For the weak version of conditional independence assumption (CIA), please see Gangl (2004)

evidence from prior studies, which found that higher-income groups who also consume more energy tend to have more scope for conservation.<sup>20</sup>

Matching Algorithms		Lower	income	Higher Income		
		Bill	kWh	Bill	kWh	
	Caliper=0.01	-0.15	-0.09	-0.24**	-0.22**	
		(0.11)	(0.09)	(0.10)	(0.09)	
Radius	Caliper=0.05	-0.17	-0.08	-0.28***	-0.24***	
Radius		(0.11)	(0.09)	(0.09)	(0.09)	
	Caliper=0.1	-0.15	-0.07	-0.26***	-0.18**	
		(0.10)	(0.08)	(0.08)	(0.08)	
	NN=1:1	-0.02	-0.05	-0.26**	-0.22*	
		(0.14)	(0.14)	(0.12)	(0.13)	
NIN	NN=1:3	-0.15	-0.07	-0.27**	-0.25**	
ININ		(0.13)	(0.12)	(0.11)	(0.10)	
	NN=1:5	-0.17	-0.10	-0.28***	-0.22**	
		(0.12)	(0.10)	(0.10)	(0.10)	
Kornol	BW=0.06	-0.16	-0.08	-0.28***	-0.24**	
Kennet		(0.10)	(0.08)	(0.10)	(0.10)	
N		531	530	494	494	

Table 5A:	Heterogeneous	impacts across	income group
-----------	---------------	----------------	--------------

Standard errors in parentheses; \* p<0.10, \*\* p<0.05, \*\*\* p<0.01

We next consider the role of education, estimating our model for subsamples differentiated based on the household head's level of education: more or less than the median number of years of schooling (eight years). Results indicate that pre-paid metering reduces electricity consumption, especially among households with relatively more educated heads (Table 5B). This is consistent with the idea that pre-paid metering may have a higher impact on households that are more aware of their electricity consumption. It should, of course, be acknowledged that such households may also have higher income and higher energy consumption (and, therefore, more potential for energy saving). The descriptive statistics also show that pre-paid meter customers are better educated than post-paid customers, although this variable is not strongly related to selection into pre-paid meters once we control for other factors (as shown in Table 3).

<sup>&</sup>lt;sup>20</sup> We have also carried out a heterogeneous impacts analysis based on the reduced sample with verified billing data in the next section discussing the IV analysis. The results are similar.

<b>Matching Algorith</b>	nms	Lower	education	Higher Education		
		Bill	kWh	Bill	kWh	
	Caliper=0.01	-0.10	0.04	-0.23***	-0.20**	
		(0.12)	(0.13)	(0.08)	(0.09)	
Padius	Caliper=0.05	-0.13	-0.05	-0.27***	-0.21***	
Radius		(0.10)	(0.12)	(0.08)	(0.08)	
	Caliper=0.1	-0.13	-0.04	-0.25***	-0.18**	
		(0.09)	(0.11)	(0.08)	(0.07)	
	NN=1:1	-0.08	-0.09	-0.25**	-0.20*	
		(0.17)	(0.15)	(0.11)	(0.11)	
NN	NN=1:3	-0.17	-0.04	-0.25**	-0.21**	
		(0.12)	(0.14)	(0.10)	(0.09)	
	NN=1:5	-0.16	-0.02	-0.25***	-0.19**	
		(0.13)	(0.12)	(0.09)	(0.08)	
Korpol	BW=0.06	-0.13	-0.04	-0.27***	-0.21**	
Kennet		(0.11)	(0.10)	(0.08)	(0.08)	
Ν		475	473	550	551	

# Table 5B: Heterogeneity across education (median education)

Standard errors in parentheses; \* p<0.10, \*\* p<0.05, \*\*\* p<0.01

Finally, stratification of sample households by meter sharing (Table 5C) shows that pre-paid metering only significantly reduces electricity consumption among households that do not share meters. This result is expected, as households that do not share meters have higher *per capita* consumption than those that do share, and we have previously shown that higher-income households reduce electricity consumption more (Table 5A). The result is also consistent with the idea that meter sharing may dilute incentives to conserve electricity, even among those with pre-paid meters, and may diminish the information content of this type of metering system.

Matching Algorithms		Do not	share meter	Share meter		
		Bill	kWh	Bill	kWh	
	Caliper=0.01	-0.22***	-0.13*	-0.04	0.02	
		(0.08)	(0.08)	(0.16)	(0.15)	
Dedius	Caliper=0.05	-0.25***	-0.16**	-0.12	-0.06	
Radius		(0.08)	(0.07)	(0.11)	(0.11)	
	Caliper=0.1	-0.23***	-0.14**	-0.11	-0.05	
		(0.08)	(0.06)	(0.11)	(0.10)	
	NN=1	-0.23*	-0.16	0.03	0.06	
		(0.12)	(0.11)	(0.15)	(0.14)	
NN	NN=3	-0.23**	-0.18*	-0.06	-0.02	
ININ		(0.10)	(0.10)	(0.14)	(0.13)	
	NN=5	-0.27***	-0.19**	-0.10	-0.04	
		(0.10)	(0.08)	(0.13)	(0.12)	
Korpol	BW=0.06	-0.25***	-0.17**	-0.12	-0.06	
Kennet		(0.07)	(0.08)	(0.13)	(0.10)	
Ν		781	780	244	244	

Standard errors in parentheses; \* p<0.10, \*\* p<0.05, \*\*\* p<0.01

## 6.3 IV estimation of treatment effects

In this section, we relax the CIA underlying PSM to allow for selection on unobservables, using the IV estimation approach described in Section 5. Before proceeding to estimate the average treatment effects, we discuss various tests of endogeneity and identification. Based on Baum *et al.* (2007), we followed these steps. First, we tested for endogeneity of pre-paid metering, using the Wu–Hausman and Durbin–Wu–Hausman tests. As presented in Appendix 5, we reject the null hypothesis that metering is an exogenous regressor for both electricity consumption and expenditure. Second, we tested for the relevance of our proposed instruments (underidentification) using Anderson's canonical correlation test. Results reject the null hypothesis that the equation is underidentified. Finally, the Sargan–Hansen test for overidentifying restrictions indicates that the instruments are valid. The results from these various tests support the use of these IVs to estimate the average impact of pre-paid metering on household electricity consumption.

The IV estimation was implemented using the probit–2SLS option of the Stata *ivtreatreg* command (Cerulli, 2014). Appendix 2 presents the covariates affecting household electricity consumption.

<sup>&</sup>lt;sup>21</sup> One of the challenges we faced was computing the actual electricity consumption of households sharing an electricity meter that is, a household's relevant electricity consumption is the total metered electricity consumed by all shared households, minus the electricity consumed by all households sharing the meter. As the survey questionnaire does not have detailed information on those sharing a meter with the surveyed household, it is impossible to fully account for their electricity consumption.

Consistent with expectations, results indicate that larger households have higher average monthly electricity consumption, as do households with higher income. The results indicate that a 10% increase in average monthly *per capita* consumption expenditure (excluding spending on electricity) is associated with 1.5% increase in electricity consumption. In addition, households with a married head of household have higher monthly electricity consumption than those with a non-married household head.

As discussed in Section 5, the IV treatment effect was estimated under two separate assumptions. First, we assume a homogeneous response to the treatment—in other words, household response to the treatment is not affected by other characteristics. The estimated average treatment effects on the treated (ATT/ATET) are somewhat higher than those obtained from PSM, indicating that hidden bias may, if anything, reduce the ATT estimated using PSM (Table 6). Pre-paid meter customers are found to have about 23% lower monthly average electricity consumption and 5% lower average monthly spending on electricity, though the latter is not significant.

# Table 6: IV-2SLS estimation of treatment effect (bootstrapped standard errors: number of replications = 100)

	ŀ	lomogen	eous trea	tment effe	ects	Heterogeneous treatment effects				
	Bill	kWh	Bill from EEU	kWh from EEU	House keeping App.	Bill	kWh	Bill from EEU	kWh from EEU	House keeping App.
Average Treatment	-0.12	-0.04	-0.05	-0.23**	-0.25***	-0.12	-0.03	- 0.05*	- 0.25**	-0.26***
Effects on the treated (ATET)	(0.09)	(0.10)	(0.03)	(0.10)	(0.10)	(0.09)	(0.09)	(0.03)	(0.10)	(0.10)
Average Treatment						-0.13	-0.05	- 0.06*	-0.19*	-0.20*
Effects on the non- treated (ATENT)						(0.10)	(0.09)	(0.03)	(0.10)	(0.11)
Average Treatment						-0.12	-0.04	- 0.06*	- 0.21**	-0.22**
Effects (ATE)						(0.09)	(0.08)	(0.03)	(0.10)	(0.10)
No. of obs	966	965	987	736	736	966	965	987	736	736

Robust standard errors in parentheses; \* p<0.10, \*\* p<0.05, \*\*\* p<0.01

The average treatment effects (with bootstrapped standard errors) obtained from the model, allowing for heterogeneous treatment effects, are similar to those with a homogeneous response. The average difference in monthly electricity consumption increases very slightly to 25% and the magnitude of the effect on monthly electricity spending is the same but significant at 10%.

Table 7 presents the estimation results, assuming heterogeneity based on income, education, and meter sharing (as we did for the PSM). The main outcome variable is electricity consumption.<sup>22</sup> In addition, we examine the effect on other outcome variables, such as the main baking stove and the number of cooking days. Results again indicate that pre-paid metering impacts differ as a function of the level of education of the household head. Specifically, pre-paid meters reduce consumption most for households with heads who have more than the median years of schooling. Pre-paid meter households with more educated heads have a 25% lower monthly average electricity consumption than their post-paid counterparts. On the other hand, households with heads with fewer than the median years of schooling have an electricity consumption that is no different from that of the post-paid group.

We completed the same analysis for the two income categories: above and below the median income. Higher-income households with pre-paid meters seem to reduce electricity consumption by around 20.5%. Among the lower-income households, we find no statistical difference between the groups (prepaid and post-paid customers), although the point estimates are similar to those in the higher-income group. The final analysis based on meter sharing shows that pre-paid meters have a larger effect on reducing electricity consumption for households that do not share meters. These results are largely consistent with those from the PSM.

	Inco	Meter S	haring	Education level		
	Below med	Above med	No	Yes	Below median	Above median
Bill from EEU	-0.27*	-0.22*	-0.25**	-0.16	-0.08	-0.29**
	(0.15)	(0.11)	(0.12)	(0.17)	(0.16)	(0.12)
Wh from EEU	-0.27	-0.23**	-0.26**	-0.17	-0.08	-0.30**
KWII HOIII EEU	(0.17)	(0.12)	(0.12)	(0.22)	(0.16)	(0.12)
Housekeeping App	-0.08*	-0.03	-0.05	-0.04	-0.05	-0.05
nousekeeping App.	(0.04)	(0.04)	(0.04)	(0.06)	(0.06)	(0.04)
Pill from survey	-0.18	-0.10	-0.16	-0.03	-0.07	-0.07
bitt from survey	(0.14)	(0.13)	(0.11)	(0.18)	(0.15)	(0.12)
kWh from survey	-0.09	-0.01	-0.07	0.03	-0.03	0.04
KWII HOIII SUIVEy	(0.15)	(0.12)	(0.10)	(0.18)	(0.16)	(0.11)
Ν	501	464	728	237	449	516

Table 7:	Heterogeneous treatment effects of pre-paid metering on electricity consumption
	(based on billing data)

Standard errors in parentheses; \* p<0.10, \*\* p<0.05, \*\*\* p<0.01

### 6.4 Summary and relation to prior estimates in the literature

<sup>&</sup>lt;sup>22</sup> We have checked whether the missing observations had any impact on our estimates. Results are shown in Appendix 3 and suggest missing data does not bias the estimation results.

Overall, these results point to significantly lower electricity consumption among households with prepaid meters relative to customers in the traditional post-paid billing system, using both PSM and IV specifications. The findings agree with previous work (for example Aliu, 2020; Ayodele *et al.*, 2017; Jack and Smith, 2020; Qiu *et al.*, 2017) showing that real-time feedback from pre-paid metering generally tends to reduce electricity consumption. Our estimates are somewhat higher than those from several previous rigorous causal impact studies (12%–13% from Jack and Smith (2020) and Qiu *et al.* (2017)) focusing on the more developed contexts of South Africa and the United States, where income constraints may bind to a lesser degree. However, one existing study from Nigeria has reported a greater difference in energy consumed (47%) between pre-paid and post-paid customers (Aliu, 2020).

In our setting, it is likely that the real-time feedback provided by pre-paid meters raises consumer consciousness of their own electricity consumption, especially among more educated households (Azila-Gbettor *et al.*, 2015; Qiu *et al.*, 2017), and allows them to understand the impacts of behaviours that reduce electricity consumption and expenditures. In addition to such real-time feedback, the nature of accounting for these costs and the timing of payment is also different for pre-paid meter users. Studies have found that pre-paid meter customers are more knowledgeable of the impacts of their electricity-related behaviours and feel they have more control over their consumption (Amnie, 2016).

Interestingly, our findings also suggest heterogeneous impacts along education, income, and meter sharing dimensions. Level of education is a major predictor of energy literacy (Brounen *et al.*, 2013)<sup>23</sup>. It may also proxy for the capacity to monitor and make rational decisions based on the information provided by pre-paid meters. Households with better education tend to conserve more energy than their counterparts. Similarly, both PSM and IV estimations indicate that pre-paid metering more significantly reduces electricity consumption among higher-income households and those who do not share meters.

Unlike other similar studies on pre-paid metering, we also examined the impact of pre-paid metering on the ownership of appliances, the type of main cooking and baking stoves, the use of energy-efficient lights, and the frequency of baking and cooking days. We further analysed whether the use of pre-paid meters affects household satisfaction with the services provided by the utility. While pre-paid metering may have a modest positive impact on the ownership of total appliances, this result is not statistically significant or fully consistent across specifications, and pre-paid billing does not appear to have a significant impact on ownership of appliances in specific categories other than energy-efficient lights. Given that education and appliance ownership may not have a clear correlation pattern (Tesfamichael *et al.*, 2021), it may not be surprising to see less of an impact of pre-paid metering on appliance ownership; moreover, while prepayment does appear to reduce consumption and expenses, increased appliance acquisition may not follow for highly budget-constrained households. Another key result of this study is that households with pre-paid meters are more satisfied with utility services. This is a new and important finding that lends support to the EEU's efforts to promote pre-paid meters.

<sup>&</sup>lt;sup>23</sup> According to Brounen et al. (2013), energy literacy means consumers' ability to calculate the long-term impacts in terms of monetary and energy savings resulting from energy efficiency investments.

# 7 Conclusion and policy implications

Pre-paid electricity metering is increasingly being adopted by many utilities and consumers around the world, but there is still relatively limited evidence of the impacts of such a system on the electricity consumption and wellbeing of households. In Ethiopia, for example, although the dissemination of pre-paid meters started more than a decade ago, its impact on electricity consumption has never been rigorously examined. This paper is an attempt to fill this gap. Using data collected from households in Addis Ababa combined with billing data obtained from the Ethiopian Electric Utility (EEU), we analysed the impact of pre-paid meters on household electricity consumption, using PSM and IV regression methods to better isolate causal impacts.

We find strong and consistent evidence that the pre-payment system significantly reduces household electricity consumption. Pre-paid meter users are found to have at least 13% lower monthly average electricity expenditure and 19% lower monthly average electricity consumption. Results are generally robust across different estimation methods, and across samples with and without verified billing data. In addition, our results indicate heterogeneous impacts along education, income, and meter sharing dimensions. Pre-paid meter households with heads who have more than the median number of years of schooling are found to reduce their electricity consumption substantially. Meanwhile, we find no significant differences between pre-paid and post-paid meter users from households with a head who has less than the median years of schooling. Greater and more consistent consumption reductions are also observed in households with more income and households that do not share meters.

Given these results, it appears that scaling up of pre-paid metering has the potential to reduce household electricity consumption substantially, and the impacts of such a change do not appear to be regressive. In other words, pre-paid metering does not have a disproportionately negative effect on consumption by lower-income households, which might be a particular concern given that such households already consumer relatively little electricity. This is vital for countries like Ethiopia, where increasing demand for electricity is a strain on existing generation capacity. Increased dissemination of pre-paid meters and the replacement of existing post-paid meters could save electricity and further enable efforts to expand electricity access, while incentivising energy conservation among higher status and consumption households. Furthermore, related empirical evidence shows that pre-paid metering can help improve cost recovery, one of the main challenges facing the EEU (Tesfamichael *et al.*, 2021). It should also be noted that pre-paid metering may be even more effective when combined with efforts to improve energy literacy. This requires the efforts of the utility company to continue educating customers about the multiple advantages of pre-paid metering.

Further research is required to understand longer-term responses to pre-paid metering, especially for outcomes such as appliance ownership, using data collected over longer periods. It is unclear whether impacts on energy consumption persist, expenditure savings continue, and what the implications of those savings are for long-term energy-using technology adoption and behaviors, and our analysis does not explore such aspects. Additional investigation is warranted to consider more carefully the behavioural adjustments made by poor households that may suffer from particularly acute energy poverty (Kambule *et al.*, 2019). Finally, the role of pre-paid metering in affecting the EEU's cost recovery efforts through addressing non-technical losses, revenue collection and other administrative burdens should be further explored.

### References

- Akele, S. (2012) 'Customer service quality in Ethiopian Electric Power Corporation (EEPCO)', Uppsala University, Department of Business Studies, pp. 1–40.
- Aliu, I.R. (2020) 'Energy efficiency in postpaid–prepaid metered homes: analyzing effects of socioeconomic, housing, and metering factors in Lagos, Nigeria', *Energy Efficiency* 13(4), pp. 853–869.
- Amnie, A. (2016) 'Assessment of implementation of power energy prepayment service system: case of Ethiopian Electric Utility (EEU)', unpublished MSc thesis, St Mary's University, Addis Ababa.
- Arawomo, D.F. (2017) 'Electricity billing systems and household electricity use behaviour in Ibadan, Nigeria', *Energy for Sustainable Development* 40, pp. 77–84.
- Arthur, M. de F.S.R., Zahran, S., and Bucini, G. (2010) 'On the adoption of electricity as a domestic source by Mozambican households', *Energy Policy* 38(11), pp. 7235–7249.
- Aydin, E., Brounen, D., and Kok, N. (2018) 'Information provision and energy consumption: evidence from a field experiment', *Energy Economics* 71, pp. 403–410.
- Ayodele, T.R., Ogunjuyigbe, A.S.O., and Atiba, I.A. (2017) 'Assessment of the impact of information feedback of prepaid meter on energy consumption of city residential buildings using bottom-up load modeling approach', *Sustainable Cities and Society*, pp. 171–183.
- Azila-Gbettor, E.M., Atatsi, E.A., and Deynu, F. (2015) 'An exploratory study of effects of prepaid metering and energy related behavior among Ghanaian households', *International Journal of Sustainable Energy and Environmental Research* 4(1), pp. 8–21.
- Bao, J., and Ho, B. (2015) 'Heterogeneous Effects of Informational Nudges on Pro-social Behavior. *BE Journal of Economic Analysis and Policy* 15(4), pp. 1619-1655.
- Baptista, I. (2015) 'We live on estimates': everyday practices of prepaid electricity and the urban condition in Maputo, Mozambique', *International Journal of Urban and Regional Research* 39(5), pp. 1004–1019.
- Baum, C.F., Schaffer, M.E., and Stillman, S. (2007) 'Enhanced routines for instrumental variables/generalized method of moments estimation and testing', *Stata Journal 7*(4), pp. 465–506.
- Becker, S.O. and Ichine, A. (2002) 'Estimation of average treatment effects based on propensity scores', *Stata Journal* 2(4), pp. 358–377.
- Blasch, J., Filippini, M., and Kumar, N. (2019) 'Boundedly rational consumers, energy and investment literacy, and the display of information on household appliances', *Resource and Energy Economics* 56, pp. 39–58.
- Blimpo, M.P. and Cosgrove-Davies, M. (2019) 'Electricity access in sub-Saharan Africa: uptake, reliability, and complementary factors for economic impact', *Africa Development Forum Series*, World Bank, Washington DC. doi:10.1596/978-1-4648-1361-0
- Brookhart, M.A., Schneeweiss, S., Rothman, K.J., Glynn, R.J., Avorn, J., and Stürmer, T. (2006) 'Variable selection for propensity score models', *American Journal of Epidemiology* 163(12), pp. 1149–1156.
- Broto, V.C., Baptista, I., Kirshner, J., Smith, S., and Neves Alves, S. (2018) 'Energy justice and sustainability transitions in Mozambique', *Applied Energy* 228, pp. 645–655.

- Brounen, D., Kok, N., and Quigley, J.M. (2013) 'Energy literacy, awareness, and conservation behavior of residential households', *Energy Economics* 38, pp. 42–50.
- Caliendo, M. and Kopeinig, S. (2008) 'Some practical guidance for the implementation of propensity score matching', *Journal of Economic Surveys* 22(1), pp. 31–72.
- Casarin, A.A. and Nicollier, L. (2009) *Prepaid Meters in Electricity. A Cost–Benefit Analysis*. IAE Working Paper Series, IAE Business School, Austral University.
- Cerulli, G. (2014) 'Ivtreatreg: a command for fitting binary treatment models with heterogeneous response to treatment and unobservable selection', *Stata Journal* 14(3), pp. 453–480.
- Cerulli, G. (2015) 'Econometric evaluation of socio-economic programs. Theory and Applications: in Advanced Studies in Theoretical and Applied Econometrics' Springer, Heidelberg New York.
- Chinomona, R. and Sandada, M. (2014) 'Customers perception on ESKOM's pre-paid billing system and the effects on their satisfaction and trust', *Mediterranean Journal of Social Sciences* 5(9), pp. 199–126.
- Colton, R.D. (2001) 'Prepayment utility meters, affordable home energy, and the low income utility consumer', *Journal of Affordable Housing & Community Development Law* 10(3), pp. 285–305.
- Darby, S. (2001) 'Making it obvious: designing feedback into energy consumption', *Energy Efficiency in Household Appliances and Lighting*, pp. 685–696.
- Dehejia, H.R. and Wahba, S. (2002) 'Score matching methods for nonexperimental causal studies', *Review of Economics and Statistics* 84, pp. 151–161.
- DiPrete, T.A. and Gangl, M. (2004) 'Assessing bias in the estimation of causal effects: Rosenbaum bounds on matching estimators and instrumental variables estimation with imperfect instruments', *Sociological Methodology* 34, pp. 271–310.
- Du, L., Guo, J., and Wei, C. (2017) 'Impact of information feedback on residential electricity demand in China', *Resources, Conservation and Recycling* 125(59), pp. 324–334.
- Esteves, G.R.T., Oliveira, F.L.C., Antunes, C.H., and Souza, R.C. (2016) 'An overview of electricity prepayment experiences and the Brazilian new regulatory framework', *Renewable and Sustainable Energy Reviews* 54, pp. 704–722.
- Fobi, S., Deshpande, V., Ondiek, S., Modi, V., and Taneja, J. (2018) 'A longitudinal study of electricity consumption growth in Kenya', *Energy Policy* 123, pp. 569–578.
- Frederick, S., Loewenstein, G., and O'Donoghue, T. (2002) 'Time discounting and time preference: a critical review', *Journal of Economic Literature* 40(2), pp. 351–401.
- Frölich, M. and Sperlich, S. (2019) *Impact Evaluation*, Cambridge University Press, Cambridge. https://doi.org/10.1017/9781107337008
- Gangl, M. (2004) 'RBOUNDS: Stata module to perform Rosenbaum sensitivity analysis for average treatment effects on the treated', Statistical Software Components S438301, Boston College Department of Economics. http://econpapers.repec.org/software/bocbocode/s438301.htm
- Getachew, E.T. (2018) 'Assessing project management practices of prepaid energy metering project in Ethiopian Electric Utility: the case of Addis Ababa', Addis Ababa University, School of Commerce, unpublished MSc thesis, Addis Ababa.

- Golumbeanu, R. and Barnes, D. (2013) 'Connection charges and electricity access in sub-Saharan Africa' [online], accessed from https://openknowledge.worldbank.org/handle/10986/15871.
- Heymans, C., Eales, K., and Franceys, R. (2014) 'The limits and possibilities of prepaid water in urban Africa: lessons from the field', World Bank Group WSP Report, Washington, DC.
- Ichino, A., Mealli, F., and Nannicini, T. (2008) 'From temporary help jobs to permanent employment : What can we learn from matching estimators and their sensitivity?' *Journal of Applied Econometrics* 23, pp. 305–327.
- Jack, B.K. and Smith, G. (2015) 'Pay as you go: prepaid metering and electricity expenditures in South Africa', *American Economic Review: Papers & Proceedings* 105(5), pp. 237–241.
- Jack, B.K. and Smith, G. (2020) 'Charging ahead: prepaid metering, electricity use, and utility revenue', *American Economic Journal: Applied Economics* 12(2), pp. 134–168.
- Jacome, V. and Ray, I. (2018) 'The prepaid electric meter: rights, relationships and reification in Unguja, Tanzania', *World Development* 105, pp. 262–272.
- Jung, J, and Liu Streeter, J.L (2015) 'Does health insurance decrease health expenditure risk in developing countries? The case of China', *Southern Economic Journal* 82(2), pp.361–384. doi:10.1002/soej.12101
- Kambule, N., Yessoufou, K., and Nwulu, N. (2018) 'A review and identification of persistent and emerging prepaid electricity meter trends', *Energy for Sustainable Development* 43, pp. 173–185.
- Kambule, N., Yessoufou, K., Nwulu, N., and Mbohwa, C. (2019) 'Temporal analysis of electricity consumption for prepaid metered low- and high-income households in Soweto, South Africa', *African Journal of Science, Technology, Innovation and Development* 11(3), pp. 375–382.
- Khandker, S.R., Koolwal, G.B., and Samad, H.A. (2010) *Handbook on Impact Evaluation: Quantitative Methods and Practices*, World Bank, Washington DC.
- Kojima, M., Zhou, X., Han, J.J., Wit, J. de, Bacon, R., and Trimble, C. (2016) 'Who uses electricity in sub-Saharan Africa?' World Bank Policy Research Working Paper 7889, August [online]. Accessed from http://documents.worldbank.org/curated/en/967821470756082684/pdf/WPS7789.pdf.
- Krishnamurti, T., Davis, A.L., Wong-Parodi, G., Wang, J., and Canfield, C. (2013) 'Creating an in-home display: Experimental evidence and guidelines for design', *Applied Energy* 108, pp. 448–458.
- Lukuyu, J., Fetter, R., Krishnapriya, P.P., Williams, N., and Taneja, J. (2021) 'Building the supply of demand: Experiments in mini-grid demand stimulation', *Development Engineering* 6, p. 100058.
- Lynham, J., Nitta, K., Saijo, T., and Tarui, N. (2016) 'Why does real-time information reduce energy consumption?' *Energy Economics* 54, pp. 173–181.
- Mahapatra, S. and Golhar, D. (2018) 'Consumers perception and satisfaction of switching from postpaid to prepaid model in electricity consumption', *International Journal of Business Management and Allied Sciences* 5(1).
- Malama, A., Mudenda, P., Ng'ombe, A., Makashini, L., and Abanda, H. (2014) 'The effects of the introduction of prepayment meters on the energy usage behaviour of different housing consumer groups in Kitwe, Zambia', *AIMS Energy* 2(3), pp. 237–259.

- Matsukawa, I. (2018) 'Information acquisition and residential electricity consumption: evidence from a field experiment', *Resource and Energy Economics* 53, pp. 1–19.
- McRae, S. (2015) 'Infrastructure quality and the subsidy trap', *American Economic Review* 105(1), pp. 35–66.
- Meles, T.H. (2020) 'Impact of power outages on households in developing countries: evidence from Ethiopia', *Energy Economics* 91, p. 104882.
- Ministry of Water, Irrigation and Electricity (MoWIE)(2019) 'Light to all, National Electrification Program 2.0: integrated planning for universal access', Ministry of Water, Irrigation and Electricity, Ethiopia.
- Muzzini, E. (2008) 'Urban poverty in Ethiopia: a multi-faceted and spatial perspective', *Urban Papers UP-4*, World Bank, Washington DC [online]. Accessed from wwwwds.worldbank.org/external/default/WDSContentServer/WDSP/IB/2008/05/29/000160016\_20080 529172824/Rendered/PDF/430270NWP0ET0u10Box327344B01PUBLIC1.pdf.
- Mwangia, A.C. and Mangusho, Y.S. (2017) 'Influence of innovation services on customer satisfaction, a case of Kenya Power and Lighting Company Limited, Elgeyo Marakwet County', *American Scientific Research Journal for Engineering, Technology, and Sciences (ASRJETS)* 37(1), pp. 273–286.
- Mwaura, F.M. (2012) 'Adopting electricity prepayment billing system to reduce non-technical energy losses in Uganda: lesson from Rwanda', *Utilities Policy* 23, pp. 72–79.
- Northeast Group (2014) 'Sub-Saharan Africa electricity metering: market forecast (2014–2024)'. Technical Report, North-east group, LLC.
- Oseni, M.O. (2015) 'Assessing the consumers' willingness to adopt a prepayment metering system in Nigeria', *Energy Policy* 86, pp. 154–165.
- O'Sullivan, K.C., Howden-Chapman, P.L., Fougere, G.M., Hales, S., and Stanley, J. (2013) 'Empowered? Examining self-disconnection in a postal survey of electricity prepayment meter consumers in New Zealand', *Energy Policy* 52, pp. 277–287.
- O'Sullivan, K.C., Stanley, J., Fougere, G., and Howden-Chapman, P. (2016) 'Heating practices and selfdisconnection among electricity prepayment meter consumers in New Zealand: a follow-up survey', *Utilities Policy* 41, pp. 139–147.
- O'Sullivan, K.C., Viggers, H.E., and Howden-Chapman, P.L. (2014) 'The influence of electricity prepayment meter use on household energy behaviour', *Sustainable Cities and Society* 13, pp. 182–191.
- O'Sullivan, K.C., Howden-Chapman, P.L., and Fougere, G.M. (2015) 'Fuel poverty, policy, and equity in New Zealand: the promise of prepayment metering', *Energy Research and Social Science* 7, pp. 99–107.
- Padam, G., Rysankova, D., Portale, E., Koo, B.B., Keller, S., and Fleurantin, G. (2018) 'Ethiopia: beyond connections', in *Energy Access Diagnostic Report Based on the Multi-Tier Framework*. https://doi.org/10.1596/24368
- Qiu, Y., Xing, B., and Wang, Y.D. (2017) 'Prepaid electricity plan and electricity consumption behavior', *Contemporary Economic Policy* 35(1), pp. 125–142.
- Ramos, A., Gago, A., Labandeira, X., and Linares, P. (2015) 'The role of information for energy efficiency in the residential sector', *Energy Economics* 52, pp. S17–S29.

- Rosenbaum, P.R. (1996) 'Observational studies and nonrandomized experiments', *Handbook of Statistics* 13, pp. 181–197.
- Rosenbaum, P.R. (2002) 'Covariance adjustment in randomized experiments and observational studies', *Statistical Science* 17(3), pp. 286–327.
- Rosenbaum, P.R. (2005) 'Sensitivity analysis in observational studies', *Encyclopedia of Statistics in Behavioral Science*. https://doi.org/10.1002/9781118445112.stat06358
- Rosenbaum, P.R. and Rubin, D.B. (1983) 'The central role of the propensity score in observational studies for causal effects', *Biometrika* 70(1), pp. 41–55.
- Rosenbaum, P.R. and Rubin, D.B. (1985) 'Constructing a control group using multivariate matched sampling methods that incorporate the propensity score', *American Statistician* 39(1), pp. 33–38.
- Rubin, D.B. (2001) 'Using propensity scores to help design observational studies: application to the tobacco litigation', *Health Services & Outcomes Research Methodology* 2, pp. 169–188.
- Schultz, P.W., Estrada, M., Schmitt, J., Sokoloski, R., and Silva-Send, N. (2015) 'Using in-home displays to provide smart meter feedback about household electricity consumption: a randomized control trial comparing kilowatts, cost, and social norms', *Energy* 90, pp. 351–358.
- Sievert, M. and Steinbuks, J. (2020) 'Willingness to pay for electricity access in extreme poverty: evidence from sub-Saharan Africa', *World Development* 128, p. 104859.
- Smith, T.B. (2004) 'Electricity theft: a comparative analysis', *Energy Policy* 32(18), pp. 2067–2076.
- Stinson, J., Willis, A., Williamson, J.B., Currie, J., and Smith, R.S. (2015) 'Visualising energy use for smart homes and informed users', *Energy Procedia* 78(0), pp. 579–584.
- Sussman, R., J. LeZaks, A. Drehobl, M. Kushler, and A. Gilleo. (2018). Examining Potential for Prepay as an Energy Efficiency Program in Minnesota. Prepared by ACEEE and Seventhwave. St. Paul: Minnesota Department of Commerce. www.cards.commerce.state.mn.us/CARDS/security/search.do?documentId=%7BF29970F3-1148-4889-9715-8C92E05F0A7F%7D
- Tesfamichael, M., Mulugetta, Y., Beyene, A.D., Sebsibie, S. (2021) 'Counting the cost: coping with tariff increases amidst power supply shortfalls in urban households in Ethiopia', *Energy Research & Social Science* 71, p. 101860.
- Tewari, D.D. and Shah, T. (2003) 'An assessment of South African prepaid electricity experiment, lessons learned, and their policy implications for developing countries', *Energy Policy* 31, pp. 911–927.
- The Allen Consulting Group (2009) 'Prepayment meter systems in Western Australia', Cost benefit analysis and regulatory impact assessment, Final report to the Economic Regulation Authority, Melbourne.
- Trimble, C., Kojima, M., Perez-Arroyo, I., and Mohammadzadeh, F. (2016) 'Financial viability of electricity sectors in sub-Saharan Africa: quasi-fiscal deficits and hidden costs', *Policy Research Working Paper 7788*, World Bank, Washington DC.
- Walta Media and Communication Corporate SC (2021) 'Ethiopian Electric Utility customer satisfaction: survey. retrieved from http://197.156.69.62/images/20210107\_EEU\_CSS\_FINAL\_English\_01.pdf.

- Wambua, A.M., Kihara, P., and Mwenemeru, H.K. (2017) 'Adoption of prepaid electricity metering system and customer satisfaction in Nairobi County, Kenya', *International Journal of Science and Research (IJSR)*, 6(9).
- Wooldridge, J.M. (2010) *Econometric Analysis of Cross Section and Panel Data*, MIT Press [online]. Available from http://books.google.com.tr/books?id=YXIxPgAACAAJ.

# Appendix 1: Matching quality indicators

Panel A: Matching quality indicator for the entire sample

Sample	Matching Algorithm	Pseudo <i>R</i> 2	LR χ2(p- value)	p>chi2	Mean standardized bias	Median Bias	Rubin's B	Rubin's R
		0.16	219.92	0.00	30.60	31.10	98.5*	1.32
Unmatched	Radius (0.01)	0.01	12.32	0.72	5.90	4.90	24.10	1.05
	Radius (0.05)	0.01	9.04	0.91	4.20	4.30	20.40	1.17
	Radius (0.1)	0.01	9.81	0.88	3.80	3.50	21.20	1.41
Matched	NN (1:1)	0.02	22.52	0.13	6.8	4.8	32.2*	1.00
	NN (1: 3)	0.01	12.06	0.74	5.1	5	23.50	1.22
	NN (1: 5)	0.01	12.16	0.73	4.7	3.7	23.60	1.15
	Kernel	0.01	9.07	0.91	4.1	4.1	20.40	1.18

Panel B: Quality of the match for the reduced sample

Sample	Sample	Pseudo R2	LR χ2(p- value)	p>chi2	Mean standardized bias	Median Bias	Rubin's B	Rubin's R
		0.19	197.00	0.00	35.40	39.30	109.7*	1.19
Unmatched	Radius (0.01)	0.01	12.21	0.73	4.70	3.40	28.5*	0.94
	Radius (0.05	0.01	10.56	0.84	5.40	5.10	26.2*	0.89
	Radius (0.1)	0.01	10.76	0.82	6.60	6.70	26.5*	1.07
	NN (1:1)	0.04	29.57	0.02	7.9	6.3	44.2*	0.89
Matched	NN (1: 3)	0.01	11.46	0.78	4.9	4.3	27.4*	1.10
	NN (1: 5)	0.01	11.65	0.77	5.4	5.1	27.6*	0.93
	Kernel	0.01	10.64	0.83	5.4	4.8	26.3*	0.90

\* if B>25%, R outside [0.5; 2]

# Appendix 2: IV-2SLS estimation of average treatment effects

Homogeneous treatment effects					;		ŀ	leterogeneous	treatment effect	s
	Bill	kWh	Bill from EEU	kWh from EEU	Housekeeping App.	Bill	kWh	Bill from EEU	kWh from EEU	Housekeeping App.
Dropaid motoring	-0.12	-0.04	-0.05*	-0.23**	-0.25**	-0.12	-0.04	-0.06*	-0.21**	-0.22**
Prepaid metering	(0.10)	(0.10)	(0.03)	(0.10)	(0.10)	(0.10)	(0.10)	(0.03)	(0.10)	(0.10)
Household size	0.10***	0.10***	0.02***	0.12***	0.12***	0.10***	0.09***	0.02***	0.12***	0.12***
Household Size	(0.02)	(0.01)	(0.00)	(0.02)	(0.02)	(0.02)	(0.01)	(0.00)	(0.02)	(0.02)
Droconco of childron under E	0.21***	0.19***	0.03*	0.09	0.09	0.22***	0.20***	0.03	0.10	0.10
Presence of children under 5	(0.06)	(0.05)	(0.02)	(0.07)	(0.07)	(0.06)	(0.06)	(0.02)	(0.07)	(0.07)
Age of bood	0.00	0.00	0.00***	0.00	0.00	0.00	0.00	0.00***	0.00	0.00
Age of field	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Say of bood	0.12*	0.09	0.05**	0.08	0.08	0.12*	0.08	0.05**	0.09	0.09
Sex of field	(0.07)	(0.07)	(0.02)	(0.08)	(0.08)	(0.07)	(0.07)	(0.02)	(0.08)	(0.08)
Schooling of hood	0.02**	0.02***	0.01***	0.01	0.01	0.01	0.01	0.01**	0.02**	0.02**
Schooling of field	(0.01)	(0.01)	(0.00)	(0.01)	(0.01)	(0.01)	(0.01)	(0.00)	(0.01)	(0.01)
	0.15***	0.15***	0.08***	0.15***	0.14***	0.18**	0.17**	0.08***	0.12*	0.12*

Consumption expenditure (ln)	(0.05)	(0.04)	(0.01)	(0.04)	(0.04)	(0.09)	(0.07)	(0.02)	(0.07)	(0.07)
Marital Status	0.22***	0.15**	0.05**	0.21**	0.21**	0.23***	0.15**	0.05**	0.20**	0.20**
Marital Status	(0.07)	(0.07)	(0.02)	(0.08)	(0.08)	(0.08)	(0.07)	(0.02)	(0.08)	(0.08)
Dwolling ownership	0.08	0.04	-0.01	0.13**	0.13**	0.09	0.05	-0.01	0.13**	0.13**
	(0.06)	(0.06)	(0.02)	(0.06)	(0.06)	(0.07)	(0.06)	(0.02)	(0.06)	(0.06)
Number of reems	0.10***	0.08***	0.04***	0.07***	0.07***	0.10***	0.09***	0.04***	0.07***	0.07***
Number of rooms	(0.02)	(0.02)	(0.01)	(0.02)	(0.02)	(0.02)	(0.02)	(0.01)	(0.02)	(0.02)
Motor charing	0.36***	0.32***	-0.03	0.37***	0.37***	0.16	0.15	-0.05*	0.27**	0.27**
Meter sharing	(0.06)	(0.06)	(0.02)	(0.07)	(0.07)	(0.12)	(0.11)	(0.03)	(0.12)	(0.12)
Hoad is wage employed	-0.05	-0.07	-0.04	-0.09	-0.09	-0.05	-0.07	-0.04	-0.06	-0.06
nead is wage employed	(0.09)	(0.08)	(0.03)	(0.09)	(0.09)	(0.09)	(0.08)	(0.03)	(0.09)	(0.09)
Hoad is colf amployed	-0.13	-0.13	0.00	0.05	0.05	-0.13	-0.13	0.00	0.07	0.06
nead is sett emptoyed	(0.09)	(0.08)	(0.03)	(0.09)	(0.09)	(0.09)	(0.08)	(0.03)	(0.09)	(0.09)
Head employed in other	-0.11	-0.13	-0.04	-0.14	-0.14	-0.12	-0.14	-0.04	-0.13	-0.13
nead employed in other	(0.08)	(0.08)	(0.02)	(0.08)	(0.08)	(0.09)	(0.08)	(0.02)	(0.08)	(0.08)
Walls made of wood and	0.27**	0.27**	-0.00	0.16	0.15	0.27**	0.27**	-0.00	0.14	0.13
mud	(0.11)	(0.12)	(0.04)	(0.13)	(0.13)	(0.12)	(0.12)	(0.04)	(0.13)	(0.13)
Walls made of concrete	0.33***	0.34***	0.07	0.27*	0.26*	0.34***	0.35***	0.07	0.26*	0.25*
walls made of concrete	(0.12)	(0.12)	(0.04)	(0.14)	(0.14)	(0.12)	(0.13)	(0.05)	(0.14)	(0.14)
Distance from nearest	0.00	0.00	0.00**	-0.00	-0.00	0.00	0.00	0.00**	-0.00	-0.00
centre	(0.00)	(0.00)	(0.00)	(0.01)	(0.01)	(0.00)	(0.00)	(0.00)	(0.01)	(0.01)
Hetero(Incons)						-0.07	-0.07	-0.01	0.09	0.08
netero(inconsy						(0.13)	(0.11)	(0.04)	(0.13)	(0.13)
Hetero(metersharing)						0.44**	0.39**	0.06	0.22	0.23
netero(inetersharing)						(0.18)	(0.17)	(0.06)	(0.21)	(0.21)
Hetero(schooling)						0.01	0.02	0.01	-0.03*	-0.03*
netero(schooting)						(0.02)	(0.02)	(0.01)	(0.02)	(0.02)
Constant	2.56***	2.73***	-0.70***	3.04***	3.10***	2.43***	2.64***	-0.67***	3.13***	3.15***
constant	(0.42)	(0.38)	(0.12)	(0.42)	(0.42)	(0.64)	(0.58)	(0.16)	(0.53)	(0.53)
No. of obs	966	965	987	736	736	966	965	987	736	736

Standard errors in parentheses \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01

# Appendix 3: Determinants of missingness

# 3A. Tabulation of missingness across metering

Missing	Prepaid metering							
missing	No	Yes	Total					
Matched	462 (77.13)	321 (70.39)	783(74.22)					
Missing	137 (22.87)	135(29.61)	272(25.78)					
Total	599	456	1055					
	100.00	100.00	100.00					

First row has *frequencies* and second row has *column percentages* 

# 3B. Determinants of missingness (logit model: status = 1 if missing)

	Model 1	Model 2
Howeshold size	0.0628	0.0616
	(1.46)	(1.22)
Children under 5	-0.384*	-0.330
	(-2.05)	(-1.64)
Are of household head	0.00887	0.0108
Age of household head	(1.38)	(1.58)
Soy of head	-0.297	-0.199
Sex of field	(-1.49)	(-0.95)
Vears of Schooling of head	0.0038	0.00021
reals of schooling of head	(0.22)	(0.01)
Per capita consumption expanditure (log)	0.138	0.148
rel capita consumption expenditure (tog)	(1.26)	(1.24)
Marital Status	-0.315	-0.309
	(-1.63)	(-1.51)
Dwelling Ownership	-0.264	-0.214
Dwetting Ownership	(-1.58)	(-1.22)
Number of rooms of dwelling	0.0121	0.0811
	(0.23)	(1.33)
Metersharing	0.411*	0.386*
hetersharing	(2.34)	(2.05)
Head is Wage employed (=1)	-0.117	-0.0563
	(-0.49)	(-0.22)

Hoad is Solf-omployed (-1)	0.0749	0.0873
neau is Self-employed (-1)	(0.30)	(0.34)
Head is Other employment type	-0.154	-0.135
head is other employment type	(-0.68)	(-0.57)
Wall of house made of wood and mud	0.497	0.507
wall of house made of wood and mud	(1.18)	(1.12)
Wall of house made of concrete	0.677	0.829
wall of house made of concrete	(1.55)	(1.78)
Distance from Utility Center	0.0694***	0.0724***
Distance from outry center	(6.57)	(6.50)
Electricity meter is prepaid	0.288	0.464**
	(1.76)	(2.65)
Monthly electricity hill(log)		0.228
Monthly electricity billiog,		(0.89)
Monthly electricity consumption in kwh (log)		-0.381
Monthly electricity consumption in kwir (tog)		(-1.46)
Appliance stock index		-0.161
		(-0.83)
Lighting appliance index		0.0483
		(0.69)
Entertainment appliance index		0.257
		(1.47)
Housekeeping appliance index		-0.223
		(-0.61)
cons	-3.420**	-3.331**
	(-3.13)	(-2.72)
N	1050	1023

*t* statistics in parentheses , p < 0.05, \*\* p < 0.01, \*\*\* p < 0.001

		r bounds		Hodges-L point es	eshmann timates	Confidence Interval (a=.95)	
Variable	Gamma	Upper Bound	Lower bound	Upper bound	Lower bound	Upper bound	Lower bound
	1.00	0.00	0.00	-0.24	-0.24	-0.32	-0.16
	1.05	0.00	0.00	-0.26	-0.23	-0.34	-0.14
	1.10	0.00	0.00	-0.28	-0.21	-0.36	-0.13
	1.15	0.00	0.00	-0.30	-0.19	-0.38	-0.11
	1.20	0.00	0.00	-0.31	-0.18	-0.39	-0.09
	1.25	0.00	0.00	-0.33	-0.16	-0.41	-0.08
Bill	1.30	0.00	0.00	-0.34	-0.15	-0.43	-0.06
	1.35	0.00	0.00	-0.36	-0.13	-0.44	-0.05
	1.40	0.00	0.00	-0.37	-0.12	-0.45	-0.03
	1.45	0.00	0.01	-0.38	-0.11	-0.47	-0.02
	1.50	0.00	0.02	-0.40	-0.09	-0.48	-0.01
	1.55	0.00	0.03	-0.41	-0.08	-0.49	0.00
	1.60	0.00	0.06	-0.42	-0.07	-0.51	0.02
	1.00	0.00	0.00	-0.16	-0.16	-0.24	-0.09
	1.05	0.00	0.00	-0.18	-0.15	-0.25	-0.08
	1.10	0.00	0.00	-0.19	-0.13	-0.27	-0.06
	1.15	0.00	0.00	-0.21	-0.12	-0.28	-0.04
kWh	1.20	0.00	0.00	-0.22	-0.10	-0.29	-0.03
	1.25	0.00	0.01	-0.24	-0.09	-0.31	-0.02
	1.30	0.00	0.02	-0.25	-0.08	-0.32	0.00
	1.35	0.00	0.04	-0.26	-0.07	-0.33	0.01
	1.40	0.00	0.08	-0.27	-0.05	-0.35	0.02
	1.00	0.00	0.00	-0.13	-0.13	-0.18	-0.08
	1.05	0.00	0.00	-0.14	-0.12	-0.19	-0.07
Entertainment	1.10	0.00	0.00	-0.15	-0.11	-0.20	-0.06
	1.15	0.00	0.00	-0.16	-0.10	-0.21	-0.05
	1.20	0.00	0.00	-0.17	-0.09	-0.22	-0.04
Appliance	1.25	0.00	0.00	-0.18	-0.08	-0.23	-0.03
	1.30	0.00	0.01	-0.19	-0.07	-0.24	-0.02
	1.35	0.00	0.01	-0.20	-0.06	-0.25	-0.01
	1.40	0.00	0.03	-0.21	-0.05	-0.25	0.00
	1.45	0.00	0.06	-0.22	-0.04	-0.26	0.01

# Appendix 4: Sensitivity analysis (Rosenbaum's bounds test): entire sample

Note: We only run the sensitivity analysis for outcomes with statistically significant average treatment effects

# B. Sensitivity analysis for the reduced sample

		r bounds		Hodges-L point es	.eshmann timates	Confidence Interval (a=.95)	
Variable	Gamma	Upper Bound	Lower bound	Upper bound	Lower bound	Upper bound	Lower bound
	1.00	0.00	0.00	-0.21	-0.21	-0.31	-0.10
	1.05	0.00	0.00	-0.23	-0.19	-0.33	-0.08
	1.10	0.00	0.00	-0.25	-0.17	-0.35	-0.06
	1.15	0.00	0.00	-0.26	-0.15	-0.37	-0.05
Bill	1.20	0.00	0.01	-0.28	-0.13	-0.39	-0.03
	1.25	0.00	0.01	-0.30	-0.12	-0.41	-0.01
	1.30	0.00	0.03	-0.31	-0.10	-0.42	0.00
	1.35	0.00	0.05	-0.33	-0.09	-0.44	0.02
	1.40	0.00	0.09	-0.34	-0.07	-0.45	0.04
	1.00	0.00	0.00	-0.22	-0.22	-0.32	-0.11
	1.05	0.00	0.00	-0.24	-0.20	-0.34	-0.09
	1.10	0.00	0.00	-0.25	-0.18	-0.36	-0.07
	1.15	0.00	0.00	-0.27	-0.16	-0.38	-0.06
kWh	1.20	0.00	0.00	-0.29	-0.14	-0.40	-0.04
	1.25	0.00	0.01	-0.31	-0.13	-0.41	-0.02
	1.30	0.00	0.02	-0.32	-0.11	-0.43	-0.01
	1.35	0.00	0.03	-0.34	-0.10	-0.45	0.01
	1.40	0.00	0.06	-0.35	-0.08	-0.46	0.02

Appendix 5:	<b>Endogeneity</b> a	nd Identification	tests
-------------	----------------------	-------------------	-------

	Bill	kWh	App. Stock	Lighting App.	Entertainment App.	Housekeeping App.	Energy Eff. bulb
Prepaid metering	-0.15	-0.06	0.12	-0.37*	-0.13**	-0.05*	0.05
	(0.10)	(0.10)	(0.08)	(0.20)	(0.06)	(0.03)	(0.05)
Controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Constant	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Ν	966	965	987	987	987	987	978
R-sq	0.22	0.21	0.38	0.07	0.33	0.26	0.03
Underidentification	303.26	302.78	304.64	304.64	304.64	304.64	306.03
p-value	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Weak identification	216.67	216.27	216.09	216.09	216.09	216.09	218.38
Overidentification	0.02	0.04	0.68	0.01	3.53	1.84	0.00
p-value	0.88	0.84	0.41	0.93	0.06	0.17	0.95
Endogeneity	0.00	0.01	0.12	1.51	3.17	4.39	0.13
p-value	0.97	0.92	0.73	0.22	0.07	0.04	0.72

# Continued

	Main cookstove	Main bakingstove	Cooking freq.	Baking freq	Satisfaction	Bill from EEU	kWh from EEU
prepaid_metering	-0.02	-0.03	-0.11	-0.07	0.02	-0.24**	-0.25**
	(0.05)	(0.05)	(0.17)	(0.10)	(0.06)	(0.11)	(0.11)
Controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Constant	Yes	Yes	Yes	Yes	Yes	Yes	Yes
="N""	987	987	987	987	987	736	736
R-sq	0.05	0.09	0.05	0.13	0.02	0.23	0.23
Underidentification	304.64	304.64	304.64	304.64	304.64	253.38	253.38
p-value	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Weak identification	216.09	216.09	216.09	216.09	216.09	188.21	188.21
Overidentification	9.60	0.97	2.58	0.08	0.95	0.11	0.08
p-value	0.00	0.33	0.11	0.78	0.33	0.74	0.78
Endogeneity	0.28	2.70	2.39	0.46	1.23	3.89	3.93
p-value	0.60	0.10	0.12	0.50	0.27	0.05	0.05

Standard errors in parentheses; \*, \*\*, \*\*\* refers to P<0.1, P<0.05, P<0.01, respectively

## About the authors

**Abebe D. Beyene:** is a senior research fellow at the Environment and Climate Research Center (ECRC) based at the Policy Studies Institute (PSI) in Ethiopia. His field of specialization is in environmental economics which include natural resource management, energy, climate change and agriculture. Methodologically he has a focus on applying micro-econometrics such as cross-section and panel data econometrics. His current research focuses on household energy choice, improved cook stove use and REDD+, forest and people's livelihood, and adaptation to climate change such as analyzing the impact of sustainable land management practices.

**Marc Jeuland:** is an Associate Professor in the Sanford School of Public Policy, with a joint appointment in the Duke Global Health Institute. His research interests include nonmarket valuation, water and sanitation, environmental health, energy poverty and transitions, trans-boundary water resource planning and management, and the impacts and economics of climate change. Jeuland's recent research includes work to understand the economic implications of climate change for water resources projects on transboundary river systems, a range of primary data collection projects related to analysis of adoption of environmental health improving technology, and analysis of the costs and benefits of environmental health interventions in developing countries.

**Samuel Sebsibie:** a researcherat Environment and Climate Research Center (ECRC), Policy Studies Institute (PSI), Ethiopia. He is currently working as a researcher in a research project on "Impacts and Drivers of Policies for Electricity Access: Micro-and-Macroeconomic evidence from Ethiopia'. His main responsibilities are project administration, work on the data management and analysis, and analyse the data together with other team members. He has also interest in applying econometrics tools in agricultural fields such as technology adoption.

**Sied Hassen** is a senior research fellow at Environment and Climate Research Center (ECRC) based at the Policy Studies Institute (PSI) in Ethiopia. He coordinates energy research program at the center. His research works emphasize on the application of microeconometrics and experimental economics to energy and agricultural economics. His current research focuses on applying econometrics tools on household and firm level energy consumption. Currently he is actively working on a project on 'Impacts and Drivers of Policies for Electricity Access: Micro-and-Macroeconomic evidence from Ethiopia'.

**Alemu Mekonnen:** is a Professor of economics at the department of Economics of Addis Ababa University and Dean of the College of Business and Economics. His research interests are on economic development and the environment. His work so far has focused on forestry, energy, climate change, and poverty and the environment. Methodologically he has a focus on non-market valuation techniques and (agricultural) household models. Particular areas of focus on Ethiopia so far include contingent valuation of community forestry, agricultural household models applied in the context of energy, choice modeling related to health and water, rural households' tree growing behavior, impacts of biofuel expansion and climate change on the economy, and poverty and the environment.

**Tensay Hadush Meles** is a postdoctoral researcher at the UCD School of Economics and UCD Energy Institute. Tensay's research interests are energy and behavioural economics, applied econometrics, environmental valuation and randomized field experiments. He applies a broad spectrum of advanced econometrics methods on research topics related to energy access and reliability, energy consumption behavior, adoption of renewable energy technologies, economic development and the environment. Currently, Tensay is working on modeling the adoption of renewable energy technologies such as heat pumps, solar photovoltaics, and electric vehicles.

**Subhrendu K. Pattanayak** is the Oak Professor of Environmental and Energy Policy at Duke University. He studies the causes and consequences of human behaviors related to the natural environment to help design and evaluate policy interventions in low-income tropical countries. His research is in three domains at the intersection of environment, development, health and energy: forest ecosystem services, environmental health (diarrhea, malaria, respiratory infections) and household energy transitions. He has focused on design of institutions and policies that are motivated by enormous inequities and a range of efficiency concerns (externalities, public goods and imperfect information and competition).

**Thomas Klug:** Experienced Research Associate and Program Coordinator with a demonstrated history of working in environmental science and international development. Skilled in Event Planning, Design, and Data Analysis. Strong research experience and passion for conservation, energy access, environmental economics, and ethics.

The views expressed in this Working Paper do not necessarily reflect the UK government's official policies.